508121595 09/14/2023

PATENT ASSIGNMENT COVER SHEET

Electronic Version v1.1 Stylesheet Version v1.2 EPAS ID: PAT8168773

SUBMISSION TYPE:	NEW ASSIGNMENT
NATURE OF CONVEYANCE:	RELEASE OF SECURITY INTEREST

CONVEYING PARTY DATA

Name	Execution Date
SILICON VALLEY BANK	05/25/2023

RECEIVING PARTY DATA

Name:	PLX OPCO INC.
Street Address:	8285 EL RIO STREET
Internal Address:	SUITE 130
City:	HOUSTON
State/Country:	TEXAS
Postal Code:	77054

PROPERTY NUMBERS Total: 3

Property Type	Number
Patent Number:	9730884
Patent Number:	9216150
Patent Number:	9226892

CORRESPONDENCE DATA

Fax Number: (650)833-2001

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent using a fax number, if provided; if that is unsuccessful, it will be sent via US Mail.

Phone: (650) 833-2000

Email: monica.moreno@us.dlapiper.com

Correspondent Name: DLA PIPER LLP (US)

Address Line 1: 2000 UNIVERSITY AVENUE

Address Line 4: EAST PALO ALTO, CALIFORNIA 94303

ATTORNEY DOCKET NUMBER:	438191.000001
NAME OF SUBMITTER:	BLAKE W. JACKSON
SIGNATURE:	/Blake W. Jackson/
DATE SIGNED:	09/14/2023

Total Attachments: 93

source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page1.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page2.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page3.tif

PATENT 508121595 REEL: 064913 FRAME: 0364

source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page4.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page5.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page6.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page7.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page8.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page9.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page10.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page11.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page12.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page13.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page14.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page15.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page16.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page17.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page18.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page19.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page20.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page21.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page22.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page23.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page24.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page25.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page26.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page27.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page28.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page29.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page30.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page31.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page32.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page33.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page34.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page35.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page36.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page37.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page38.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page39.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page40.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page41.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page42.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page43.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page44.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page45.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page46.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page47.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page48.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page49.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page50.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page51.tif

source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page52.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page53.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page54.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page55.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page56.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page57.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page58.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page59.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page60.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page61.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page62.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page63.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page64.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page65.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page66.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page67.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page68.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page69.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page70.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page71.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page72.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page73.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page74.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page75.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page76.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page77.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page78.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page79.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page80.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page81.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page82.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page83.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page84.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page85.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page86.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page87.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page88.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page89.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page90.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page91.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page92.tif source=PLx - [DI 147] Sale Order Approving MPG Stalking Horse APA (entered 5-25-2023)#page93.tif

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:	Chapter 1	1
	-	

PLX PHARMA INC., et al., 1 Case No. 23-10456 (MFW)

Debtors. (Jointly Administered)

Re: D.I. 19, 53 and 101

ORDER (A) APPROVING AND AUTHORIZING THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS TO PLX ACQUISITION COMPANY, LLC FREE AND CLEAR OF ALL LIENS, CLAIMS, ENCUMBRANCES AND OTHER INTERESTS, (B) APPROVING THE ASSUMPTION AND ASSIGNMENT OF CERTAIN EXECUTORY CONTRACTS AND (C) GRANTING RELATED RELIEF

Upon the motion [Docket No. 19] (as supplemented at Docket No. 53, the "Sale Motion")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") for the entry of an order (this "Sale Order"), pursuant to sections 105(a), 363 and 365 of title 11 of the United States Code (as amended from time to time, the "Bankruptcy Code"), Rules 2002, 6004, and 6006 of the Federal Rules of Bankruptcy Procedure (as amended from time to time, the "Bankruptcy Rules"), and Rule 6004-1 of the Local Rules of Bankruptcy Practice and Procedure of the United States Bankruptcy Court for the District of Delaware (as amended from time to time, the "Local Rules"), (a) authorizing the sale of substantially all of the Debtors' assets free and clear of all liens, claims and encumbrances to PLx Acquisition Company, LLC (together with Greenwood and

30280469.4

¹ The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, are: PLx Pharma Inc. (5704) and PLx Opco Inc. (6588). The mailing address for each of the Debtors is 9 Fishers Lane, Suite E, Sparta, NJ 07871.

² Debtors' Motion For Entry of (A) Order (I) Approving Bidding Procedures in Connection With Sale of Assets of the Debtors and Related Bid Protections, (II) Approving form and manner of notice, (III) Scheduling Auction and Sale Hearing, (IV) Authorizing Procedures Governing Assumption and Assignment of Certain Contracts and Unexpired Leases, and (V) Granting Related Relief; and (B) Order (I) Approving Asset Purchase Agreement, and (II) Authorizing Sale Free and Clear of All Liens, Claims, Encumbrances, and Other Interests [D.1. 19] (the "Sale Motion"). All capitalized terms used but not otherwise defined herein shall have the meaning ascribed to such term in the Sale Motion or the Stalking Horse APA, as applicable.

MPG, the "Stalking Horse Bidder"), pursuant to the terms and conditions of the Asset Purchase Agreement between the Debtors and the Stalking Horse Bidder attached hereto as Exhibit A (as amended, the "Stalking Horse APA"); (b) authorizing and approving the closing under, and the terms and conditions of, the Stalking Horse APA (including all actions taken or required to be taken in connection with the implementation and consummation of the Stalking Horse APA); (c) authorizing and approving the assumption and assignment of the Assumed Contracts in connection with the Sale; and (d) granting certain related relief, all as more fully described in the Sale Motion and Stalking Horse APA; and the Court having entered an order on May 10, 2023 [D.I. 101] (the "Bidding Procedures Order") (x) approving procedures (the "Bidding Procedures") for (i) submitting bids for the purchase of the Acquired Assets and, if necessary, (ii) conducting an auction for the Acquired Assets (the "Auction"), and (y) approving procedures for the assumption and assignment of executory contracts in connection with the Sale; and the Court having reviewed the Sale Motion and the Supporting Declarations (as defined below) and the Court having found that (i) this Court has jurisdiction over the Debtors, their estates, property of their estates and to consider the Sale Motion and the relief requested therein under 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012 (Sleet, C.J.), (ii) this Court may enter a final order consistent with Article III of the United States Constitution, (iii) venue is proper in this District pursuant to 28 U.S.C. §§ 1408 and 1409, (iv) this is a core proceeding pursuant to 28 U.S.C. § 157(b), and (v) the Debtors' notice of the Sale Motion and the assumption and assignment of the Assumed Contracts in connection with the Sale and the opportunity for a hearing on such matters were adequate and appropriate under the circumstances and no other or further notice thereof need be provided; and after due deliberation; and the Court having conducted a hearing on the Sale Motion

30280469.4 - 2 -

on May 25, 2023 (the "Sale Hearing") and having determined that the relief requested in the Sale

Motion is in the best interests of the Debtors, their estates, their creditors and other parties in

interest; and good and sufficient cause having been shown;

IT IS FOUND AND DETERMINED THAT:

A. The findings of fact and conclusions of law set forth herein constitute the Court's

findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this

proceeding pursuant to Bankruptcy Rule 9014. The findings made herein are based on the

evidence and proffers presented at or in connection with the Sale Hearing, including the Supporting

Declarations (as defined below).

B. To the extent any of the following findings of fact constitute conclusions of law,

they are hereby adopted as such. To the extent any of the following conclusions of law constitute

findings of fact, they are hereby adopted as such. Any findings of fact or conclusions of law stated

by the Court on the record at the Sale Hearing are hereby incorporated, to the extent they are not

inconsistent herewith.

I. <u>Jurisdiction, Final Order and Statutory Predicates</u>

C. The Court has jurisdiction over this matter, over the property of the Debtors and

their bankruptcy estates, including the Acquired Assets, pursuant to 28 U.S.C. §§ 157(a) and 1334.

The Court has jurisdiction to hear and determine the Sale Motion pursuant to 28 U.S.C. §§

157(b)(1) and 1334(a). This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(A), (N) and

(O). Venue of the Chapter 11 Cases and the Sale Motion is proper in this District and in the Court

pursuant to 28 U.S.C. §§ 1408 and 1409.

D. This Sale Order constitutes a final and appealable order within the meaning of 28

U.S.C. § 158(a). Notwithstanding Bankruptcy Rules 6004(h) and 6006(d), and to any extent

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 4 of 30

necessary under Bankruptcy Rule 9014 and Rule 54(b) of the Federal Rules of Civil Procedure, as

made applicable by Bankruptcy Rule 7054, the Court expressly finds that cause exists to waive all

applicable stays with respect to the effectiveness of this Sale Order, which shall be effective

immediately upon entry on the docket.

E. The statutory predicates for the relief requested in the Sale Motion are sections

105(a), 363 and 365 of the Bankruptcy Code, Bankruptcy Rules 2002, 6004, and 6006 and Local

Rule 6004-1.

II. Compliance with Bidding Procedures Order

F. As demonstrated by (i) the testimony and other evidence proffered in connection

with the Sale Motion, including, without limitation, that certain (a) Declaration of Simon Wein

[D.I. 20], (b) Supplemental Declaration of Simon Wein [D.I. 87] and (c) Declaration of John

Halloran [D.I. 137] (collectively, the "Supporting Declarations"), and (ii) the representations of

counsel made on the record at the Sale Hearing, the Debtors and their professionals have fully and

fairly marketed the Acquired Assets and conducted the sale process in compliance with the

Bidding Procedures and the Bidding Procedures Order. All interested persons and entities have

been afforded a full, fair and reasonable opportunity to (i) conduct due diligence investigations,

(ii) submit bids to purchase the Acquired Assets, and (iii) object or be heard with respect to the

Sale Motion and the relief granted by this Sale Order, including, without limitation, with respect

to the assumption and assignment of the Assumed Contracts in connection with the Sale. The

Bidding Procedures were non-collusive, formulated and implemented in good faith, were

substantively and procedurally fair to all parties, and obtained the highest or otherwise best value

for the Acquired Assets for the Debtors, their estates, and creditors under the circumstances.

30280469.4 - 4 -

G. The Stalking Horse Bidder is the Successful Bidder for the Acquired Assets in accordance with the Bidding Procedures Order, and the consideration provided by the Stalking Horse Bidder under the Stalking Horse APA constitutes the highest or otherwise best offer, and provides fair and reasonable consideration to the Debtors and their estates for the sale of the

Acquired Assets and the assumption and assignment of the Assumed Contracts.

III. Notice of the Sale, Auction and the Cure Costs

H. As evidenced by the related certificates of service, actual written notice of the Sale

Hearing, the Auction, the Sale Motion, the Sale, and the assumption and assignment of the

Assumed Contracts and the various dates and deadlines related thereto, and a reasonable

opportunity to object or be heard with respect to such matters, including, without limitation, the

Sale Motion and the relief requested therein, has been afforded to all known interested persons and

entities, including, but not limited to, the following parties: (i) the Office of the United States

Trustee for the District of Delaware; (ii) counsel to the Stalking Horse Bidder; (iii) the Internal

Revenue Service; (iv) the United States Attorney's Office for the District of Delaware; (v) all

parties that have requested notice in the Chapter 11 Cases pursuant to Bankruptcy Rule 2002; (vi)

all persons or entities known to the Debtors that have or have asserted a lien on, or security interest

in, all or any portion of the Acquired Assets; (vii) all non-Debtor parties to each Assumed Contract

(each a "Contract Counterparty" and collectively, the "Contract Counterparties"); (viii) all

potential bidders previously identified or otherwise known to the Debtors; and (ix) all creditors of

the Debtors, including those reflected on the Debtors' creditor matrix filed on the docket of the

Chapter 11 Cases.

I. As evidenced by the certificates of service filed with the Court, proper, timely,

adequate, and sufficient notice of the Sale Motion, the Auction, the Sale Hearing, and the Sale of

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 6 of 30

the Acquired Assets, including the assumption and assignment of the Assumed Contracts, has been

provided in accordance with sections 102(1), 363 and 365 of the Bankruptcy Code, Bankruptcy

Rules 2002, 6004, 6006 and 9014, and all applicable Local Rules. The notices described above

were good, sufficient, and appropriate under the circumstances. No other or further notice of the

Sale Motion, the Auction, the Sale Hearing, the Sale, or the assumption, assignment and sale of

the Assumed Contracts is required, and due process has been satisfied.

J. In accordance with the provisions of the Bidding Procedures Order, as evidenced

by the certificates of service on the docket, the Debtors served notice (the "Cure Notice") upon the

Stalking Horse Bidder and the Contract Counterparties: (i) that the Debtors seek to potentially

assume and assign the Assumed Contracts on the Closing Date; (ii) of the relevant Cure Costs; and

(iii) the deadline by which any Contract Counterparty must object to the potential assignment and

assumption of the Assumed Contracts. Pursuant to Bankruptcy Rule 6006(c), the service of such

Cure Notice was good, sufficient and appropriate under the circumstances, and no further notice

need be given in respect of establishing a cure amount for the Assumed Contracts or regarding the

potential assignment and assumption of the Assumed Contracts. Each Contract Counterparty has

had a full and fair opportunity to object to the Cure Costs set forth in the Cure Notice and to the

potential assumption and assignment of the Assumed Contracts. Other than the Eversana Cure

Objection (as defined below), which is resolved as provided for in this Sale Order, no objections

were filed with this Court with respect to the Cure Notice.

K. On May 19, 2023, in accordance with the Bidding Procedures Order, the Debtors

cancelled the Auction because the Debtors did not receive any Qualified Bids other than the

Stalking Horse Bid, and on the same day, filed the *Notice of Canceled Auction* [D.I. 121].

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 7 of 30

L. On May 23, 2023, the Debtors filed and served a notice of the revised proposed

Sale Order, an amendment to the Stalking Horse APA, and the designation of the Stalking Horse

Bidder as the Successful Bidder [D.I. 125].

M. For the sake of clarity, paragraph 9 of the Bidding Procedures Order modified terms

of the Stalking Horse APA.

IV. <u>Highest or Best Offer</u>

N. The Debtors solicited offers in accordance with the provisions of the Bidding

Procedures Order. In accordance with the Bidding Procedures Order, all persons and entities were

afforded a full, fair and reasonable opportunity to make a higher or otherwise better offer to

purchase the Acquired Assets, as compared to the offer pursuant to the Stalking Horse APA (the

"Stalking Horse Bid"). The marketing and bidding processes implemented by the Debtors and

their advisors, as set forth in the Sale Motion, were fair, proper, and reasonably calculated to result

in the best value received for the Acquired Assets. Other than the Stalking Horse Bid, no other

Qualified Bids were received prior to the Bid Deadline.

O. The Stalking Horse APA constitutes the highest or best offer for the Acquired

Assets, is fair and reasonable, and will provide a greater recovery for the Debtors' estates than

would be provided by any other available alternative. The Debtors' determination that the Stalking

Horse APA constitutes the highest or best offer for the Acquired Assets constitutes a valid and

sound exercise of the Debtors' business judgment.

P. The consideration provided by the Stalking Horse Bidder pursuant to the Stalking

Horse APA is fair and adequate and constitutes reasonably equivalent value, fair consideration,

and reasonable, current market value for the Acquired Assets under the Bankruptcy Code, the

Uniform Fraudulent Transfer Act, the Uniform Voidable Conveyance Act, and under any other

30280469.4 - 7 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 8 of 30

applicable laws of the United States, any state, territory, possession, or the District of Columbia.

Q. Approval of the Sale Motion and the Stalking Horse APA and the consummation

of the transactions contemplated thereby is in the best interests of the Debtors, their creditors, their

estates, and other parties in interest.

R. The Debtors have demonstrated compelling circumstances and a good, sufficient,

and sound business purpose and justification to enter into the Stalking Horse APA, sell the

Acquired Assets, and assume and assign the Assumed Contracts under Sections 363 and 365 of

the Bankruptcy Code, and such actions are appropriate and reasonable exercises of the Debtors'

business judgment and in the best interests of the Debtors, their estates, their creditors, and all

other parties in interest. Such business reasons include, but are not limited to, the facts that (i) there

is a substantial risk of deterioration of the value of the Acquired Assets if the Sale is not

consummated as soon as possible; (ii) the Stalking Horse APA constitutes the highest or otherwise

best offer for the Acquired Assets; (iii) no other person or entity has offered to purchase the

Acquired Assets for greater economic value to the Debtors' estates than the Stalking Horse Bidder;

and (iv) the Sale presents the best opportunity to realize the value of the Debtors' assets. In

addition, the Debtors' determination that the Stalking Horse APA constitutes the highest or

otherwise best offer for the Acquired Assets constitutes a valid and sound exercise of the Debtors'

business judgment.

V. Credit Bid

S. Pursuant to the Stalking Horse APA, the Bidding Procedures Order, and as

authorized by sections 363(b) and 363(k) of the Bankruptcy Code, the Stalking Horse Bidder is

entitled to credit bid on the Credit Bid Inventory (as defined in the Bidding Procedures Order) up

to the amount of the Allowed MPG Claim (as defined in the Bidding Procedures Order). Any

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 9 of 30

credit bid made by the Stalking Horse Bidder is a valid and proper offer pursuant to the Bidding

Procedures Order and Bankruptcy Code sections 363(b) and 363(k).

T. The reconciled amount of the Remaining Stalking Horse Bidder Claim (as defined

below) is \$1,700,000.00. Pursuant to paragraph 31(D) of the Bidding Procedures Order, the

reconciled amount of the Stalking Horse Bidder's allowed, secured Claim and the Remaining

Stalking Horse Bidder Claim were made in consultation with Patheon Pharmaceuticals, Inc. and

Patheon Manufacturing Services LLC (together, "Patheon").

VI. Authority to Consummate Sale

U. The Debtors are the sole and lawful owner of, and have clear and marketable title

to, the Acquired Assets to be sold pursuant to the Stalking Horse APA. The Debtors' right, title,

and interest in and to the Acquired Assets constitute property of the Sellers' estates and title thereto

is vested in the Sellers' estates within the meaning of section 541(a) of the Bankruptcy Code.

V. The Debtors have full corporate power and authority to execute and deliver the

Stalking Horse APA and the Transaction Documents, to perform their obligations therein, and to

consummate the Sale, and the Sale has been duly and validly authorized by all necessary corporate

action of the Debtors. No further consents or approvals are required for the Debtors to consummate

the Sale contemplated by the Stalking Horse APA.

W. The transfer of the Acquired Assets to the Stalking Horse Bidder will be, as of the

closing of the Sale (the "Closing Date"), a legal, valid, and effective transfer of such assets, and

vests or will vest the Stalking Horse Bidder with all rights, title, and interests of the Debtors to the

Acquired Assets, free and clear of all liens, claims, and encumbrances, except for the Permitted

Liens, if any, and the Assumed Liabilities (as each such term is defined in the Stalking Horse

APA), with all such liens, claims, and encumbrances to attach to any cash proceeds of the Sale

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 10 of 30

received by the Debtors with the same priority, validity, force and effect such liens, claims, and

encumbrances had in the Acquired Assets as of the Closing Date, subject to any rights, claims and

defenses the Debtors and their estates may possess with respect thereto.

VII. Good Faith of Buyer

X. The Stalking Horse Bidder is purchasing the Acquired Assets in good faith and is

a good faith buyer within the meaning of section 363(m) of the Bankruptcy Code, and is therefore

entitled to the full protection of that provision, and otherwise has proceeded in good faith in all

respects in connection with this proceeding in that, inter alia: (a) the Stalking Horse Bidder

recognized that the Debtors were free to deal with any other party interested in acquiring the

Acquired Assets; (b) the Stalking Horse Bidder complied with the provisions in the Bidding

Procedures Order; (c) the Stalking Horse Bidder agreed to subject its bid to the competitive bidding

procedures set forth in the Bidding Procedures Order; (d) the Stalking Horse Bidder in no way

induced or caused the chapter 11 filings by the Debtors; (e) all payments to be made by the Stalking

Horse Bidder and other agreements or arrangements entered into by the Stalking Horse Bidder in

connection with the Sale have been disclosed; (f) the Stalking Horse Bidder has not violated

section 363(n) of the Bankruptcy Code by any action or inaction; (g) no common identity of

directors or controlling stockholders exists between the Stalking Horse Bidder and the Debtors;

(h) the Stalking Horse Bidder is not an "insider" of the Debtors, as that term is defined in section

101(31) of the Bankruptcy Code; and (i) the Stalking Horse APA, and any other agreements or

instruments related thereto, were negotiated and executed at arm's length, without collusion and

in good faith.

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 11 of 30

VIII. No Fraudulent Transfer; Not a Successor

Y. The consideration provided by the Stalking Horse Bidder for the Acquired Assets

pursuant to the Stalking Horse APA: (i) is the highest or otherwise best offer for the Acquired

Assets; (ii) will provide a greater recovery for the Debtors' estates and creditors than would be

provided by any other practical available alternative; and (iii) constitutes reasonably equivalent

value and fair consideration for the Acquired Assets.

Z. No sale, transfer or other disposition of the Acquired Assets pursuant to the Stalking

Horse APA, or entry into the Stalking Horse APA, will subject the Stalking Horse Bidder to any

liability for claims, obligations, or encumbrances asserted against the Debtors or the Debtors'

interests in the Acquired Assets—other than the Permitted Liens, if any, and the Assumed

Liabilities as and to the extent set forth in the Stalking Horse APA—by reason of such transfer

under any laws, including any bulk-transfer laws or any theory of successor or transferee liability,

antitrust, environmental, product line, de facto merger, or substantial continuity or similar theories.

By virtue of the consummation of the transactions contemplated under the Stalking Horse APA,

(i) the Stalking Horse Bidder is not a mere continuation of the Debtors or their estates and there is

no continuity of enterprise between the Stalking Horse Bidder and the Debtors, (ii) the Stalking

Horse Bidder is not holding itself out to the public as a continuation of the Debtors or their estates,

including, but not limited to, any warranty or similar obligations, and (iii) the Sale does not amount

to a consolidation, merger, or de facto merger of the Stalking Horse Bidder and the Debtors or the

Debtors' estates. Accordingly, the Stalking Horse Bidder is not and shall not be deemed a

successor to the Debtors or their estates by reason of any theory of law or equity, and, except with

respect to the Permitted Liens, if any, and the Assumed Liabilities, the Stalking Horse Bidder's

acquisition of the Acquired Assets shall be free and clear of any "successor liability" claims of any

30280469.4 - 11 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 12 of 30

nature whatsoever. The Stalking Horse Bidder would not acquire the Acquired Assets but for the

protections against any claims based upon "successor liability" theories.

IX. Section 363(f) Is Satisfied

> Expect as expressly provided in the Stalking Horse APA or this Sale Order, to the AA.

fullest extent permitted under the Bankruptcy Code, the Acquired Assets shall be sold free and

clear of all liens (including, without limitation, mechanics' materialmens' and other consensual

and non-consensual liens and statutory liens), claims (including any "claim" as defined in section

101(5) of the Bankruptcy Code), encumbrances, interests, obligations, rights, pledges, mortgages,

deeds of trust, security interests, liabilities, debt obligations, penalties, charges, judgments,

guaranties, and indemnity obligations, in each case, of whatever kind, nature, or description in,

against, or with respect to any of the Acquired Assets, having arisen, existed, or accrued prior to

the Closing Date, whether direct or indirect, absolute or contingent, choate or inchoate, filed or

unfiled, scheduled or unscheduled, noticed or unnoticed, recorded or unrecorded, perfected or

unperfected, material or non-material, disputed or undisputed, known or unknown, matured or

unmatured, liquidated or unliquidated, arising or imposed by agreement, understanding, law,

equity, statute, or otherwise, including claims or liabilities arising under doctrines of successor

liability, or de facto merger (other than the Permitted Liens, if any, and the Assumed Liabilities,

collectively, the "Interests"), with all such Interests to attach to any cash proceeds of the Sale

received by the Debtors with the same priority, validity, force and effect such Interests had in the

Acquired Assets as of the Closing Date, subject to any rights, claims and defenses the Debtors and

their estates may possess with respect thereto.

BB. The Stalking Horse Bidder would not have entered into the Stalking Horse APA

and would not consummate the transactions contemplated thereby (by paying the Purchase Price

- 12 -30280469.4

REEL: 064913 FRAME: 0378

and assuming the Assumed Liabilities) if the sale of the Acquired Assets to the Stalking Horse

Bidder, and the assumption, assignment and sale of the Assumed Contracts to the Stalking Horse

Bidder, were not free and clear of all Interests, or if the Stalking Horse Bidder, its affiliates, or

their respective officers, directors or shareholders, or the Acquired Assets, would, or in the future

could be liable for any Interests, or could or would be required to satisfy in any manner, whether

at law or equity, or by payment, setoff, or otherwise, directly or indirectly, any Interest.

CC. Not selling the Acquired Assets free and clear of any and all Interests would

adversely impact the Debtors' estates, and the Sale of the Acquired Assets other than as free and

clear of all Interests would be of substantially less value to the Debtors' estates.

DD. The Debtors may sell the Acquired Assets free and clear of all Interests because, in

each case, one or more of the standards set forth in section 363(f)(1)-(5) of the Bankruptcy Code

has been satisfied. Those holders of Interests against the Debtors, their estates or any of the

Acquired Assets who did not object, or who withdrew their objections, to the Sale or the Sale

Motion are deemed to have consented to the Sale pursuant to section 363(f)(2) of the Bankruptcy

Code. All other holders of Interests, including whose who maintained and did not withdraw

objections to the Sale or the Sale Motion, if any, fall within one or more of the other subsections

of section 363(f) of the Bankruptcy Code.

X. Validity of Transfer

EE. The consummation of the Stalking Horse APA and the Sale is legal, valid and

properly authorized under all applicable provisions of the Bankruptcy Code, including, without

limitation, Sections 105(a), 363(b), 363(f), 363(m), 365(b), and 365(f) of the Bankruptcy Code,

and all of the applicable requirements of such sections have been complied with in respect of the

Sale.

30280469.4

FF. The transfer of the Acquired Assets to the Stalking Horse Bidder under the Stalking

Horse APA will be a legal, valid and effective transfer of all of the legal, equitable and beneficial

right, title and interest of the Debtors in and to the Acquired Assets free and clear of all Interests

as provided for herein.

XI. Assumption and Assignment of the Assumed Contracts

GG. The assumption and assignment of the executory contracts and/or unexpired leases

listed in the Stalking Horse APA (the "Assumed Contracts") pursuant to the terms of this Sale

Order is integral to the Stalking Horse APA and is in the best interests of the Debtors and their

estates, their creditors, and all other parties in interest, and represents the reasonable exercise of

sound and prudent business judgment by the Debtors. Subject to the terms and conditions of the

Stalking Horse APA, the Stalking Horse Bidder shall: (a) to the extent necessary, cure or provide

adequate assurance of cure, of any default existing prior to the date hereof with respect to the

Assumed Contracts, within the meaning of sections 365(b)(1)(A) and 365(f)(2)(A) of the

Bankruptcy Code, and (b) to the extent necessary, provide adequate assurance to any party for any

actual pecuniary loss to such party resulting from a default prior to the date hereof with respect to

the Assumed Contracts, within the meaning of sections 365(b)(1)(B) and 365(f)(2)(A) of the

Bankruptcy Code. The Stalking Horse Bidder's obligation under the Stalking Horse APA to pay

the Cure Costs and to perform the obligations under the Assumed Contracts shall constitute

adequate assurance of future performance within the meaning of sections 365(b)(1)(C) and

365(f)(2)(B) of the Bankruptcy Code.

XII. Compelling Circumstances for an Immediate Sale

HH. To maximize the value of the Acquired Assets, preserve the viability of the

Business, and avoid deterioration, erosion of value, and/or uncertainty with respect to the future

30280469.4 - 14 -

operation of the Acquired Assets, the Sale of the Acquired Assets must occur as soon as possible.

Accordingly, there is cause to waive the stays contemplated by Bankruptcy Rules 6004 and 6006.

II. The Debtors have demonstrated compelling circumstances and good, sufficient,

and sound business purpose and justification for the approval and consummation of the Sale prior

to, and outside of, a chapter 11 plan. The Sale pursuant to the Stalking Horse APA neither

impermissibly restructures the rights of the Debtors' creditors nor impermissibly dictates the terms

of a liquidating chapter 11 plan for the Debtors and, therefore, does not constitute a *sub rosa* plan.

NOW, THEREFORE, IT IS HEREBY ORDERED, ADJUDGED AND DECREED THAT:

I. General Provisions

1. The relief requested in the Sale Motion is granted and approved, and the Sale

contemplated thereby is approved as set forth in this Sale Order.

2. This Court's findings of fact and conclusions of law, set forth in the Bidding

Procedures Order, are incorporated herein by reference.

3. All objections to the Sale Motion or the relief requested therein that have not been

withdrawn, waived, or settled, and all reservations of rights included therein, are hereby overruled

on the merits and denied with prejudice. All persons and entities given notice of the Sale Motion

that failed to timely object thereto, including, without limitation, all Contract Counterparties, are

deemed to have consented to the relief sought therein, including, but not limited to, the Cure Costs

set forth in the Cure Notice.

II. Approval of the Stalking Horse APA

4. The Stalking Horse APA and all other ancillary documents, including the

Transaction Documents (as defined in the Stalking Horse APA) and all of the terms and conditions

thereof, and the Debtors' entry into such documents, are hereby approved in all respects.

30280469.4 - 15 -

empowered to take any and all actions necessary or appropriate to (a) consummate the Sale of the Acquired Assets to the Stalking Horse Bidder pursuant to and in accordance with the terms and conditions of the Stalking Horse APA, (b) close the Sale as contemplated in the Stalking Horse APA and this Sale Order, (c) execute and deliver, perform under, consummate, implement and close fully the Stalking Horse APA, together with all additional instruments and documents that may be reasonably necessary or desirable to implement the Stalking Horse APA and the Sale, including any other Transaction Documents, or as may be reasonably necessary or appropriate to the performance of the obligations as contemplated by the Stalking Horse APA and such other Transaction Documents, (d) assume, assign and transfer the Assumed Contracts, without further application to or order of this Court and upon the terms and conditions of the Stalking Horse APA, (e) perform any remaining obligations specified in the Stalking Horse APA after the Closing, and (f) otherwise implement and provide the Stalking Horse Bidder and the Debtors with the full benefit of the transactions contemplated by the Stalking Horse APA and this Sale Order.

III. Transfer of the Acquired Assets

6. Pursuant to sections 105(a), 363(b), 363(f), 365(b) and 365(f) of the Bankruptcy Code, the Debtors are authorized to transfer the Acquired Assets on the Closing Date in accordance with the terms of the Stalking Horse APA. Such Acquired Assets shall be transferred to the Stalking Horse Bidder upon and as of the Closing, and such transfer shall constitute a legal, valid, binding and effective transfer of the Acquired Assets which shall vest the Stalking Horse Bidder with all right, title and interest of the Debtors in and to the Acquired Assets free and clear of all Interests, except the Permitted Liens, if any, and the Assumed Liabilities. Upon the Closing, the

30280469.4 - 16 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 17 of 30

Stalking Horse Bidder shall take title to and possession of the Acquired Assets, subject only to the

Permitted Liens, if any, and the Assumed Liabilities.

7. Except as expressly provided by the Stalking Horse APA with respect to Permitted

Liens, if any, and Assumed Liabilities, all persons and entities (including, but not limited to, all

noteholders, warrant holders, equity security holders, affiliates, governmental, tax and regulatory

authorities, lenders, current or former customers, vendors, employees, trade creditors, litigation

claimants, judgment creditors, and other creditors) holding or asserting Interests in all or any

portion of the Acquired Assets and/or arising under or out of, in connection with, or in any way

relating to the Debtors, the Acquired Assets, the operation of the Debtors' Business prior to the

Closing Date or the transfer of the Acquired Assets to the Stalking Horse Bidder are hereby forever

barred, estopped and permanently enjoined from asserting against the Stalking Horse Bidder or its

successors or assigns, its property or the Acquired Assets, such persons' or entities' Interests.

Following the Closing, no holder of any Interest shall interfere with the Stalking Horse Bidder's

title to or use and enjoyment of the Acquired Assets based on or related to any such Interest, or

based on any action the Debtors may take in the Chapter 11 Cases.

8. All persons and entities are hereby forever prohibited and enjoined from taking any

action that would adversely affect or interfere with the ability of the Debtors to sell and transfer

the Acquired Assets to the Stalking Horse Bidder in accordance with the terms of the Stalking

Horse APA and this Sale Order.

9. All persons and entities that are in possession of some or all of the Acquired Assets

on the Closing Date are directed to surrender possession of such Acquired Assets to the Stalking

Horse Bidder at the Closing. If there is any dispute as to what constitutes an Acquired Asset, the

Court shall retain jurisdiction to resolve such dispute.

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 18 of 30

10. Upon Closing, this Sale Order shall be construed as, and shall constitute for any

and all purposes, a full and complete general assignment, conveyance and transfer of the Acquired

Assets pursuant to the terms of the Stalking Horse APA.

11. A certified copy of this Sale Order may be filed with the appropriate clerk and/or

recorded with the recorder to act to cancel any of the liens, claims, and encumbrances of record.

12. If any person or entity which has filed statements or other documents or agreements

evidencing Interests with regard to all or any portion of the Acquired Assets shall not have

delivered to the Debtors prior to the Closing, in proper form for filing and executed by the

appropriate parties, termination statements, instruments of satisfaction, releases of liens and

easements, and any other documents necessary or desirable to the Stalking Horse Bidder for the

purpose of documenting the release of all Interests, which the person or entity has or may assert

with respect to all or any portion of the Acquired Assets, the Debtors are hereby authorized,

empowered and directed, and the Stalking Horse Bidder is hereby authorized and empowered, to

execute and file such statements, instruments, releases and other documents on behalf of and in

the name of such person or entity, as attorney-in-fact, with respect to the Acquired Assets.

13. This Sale Order is and shall be binding upon and govern the acts of all persons and

entities, including, without limitation, all filing agents, filing officers, title agents, title companies,

recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies,

governmental departments, secretaries of state, federal and local officials, and all other persons

and entities who may be required by operation of law, the duties of their office, or contract, to

accept, file, register or otherwise record or release any documents or instruments, or who may be

required to report or insure any title or state of title in or to any lease; and each of the foregoing

persons and entities is hereby directed to accept for filing any and all of the documents and

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 19 of 30

instruments necessary and appropriate to consummate the transactions contemplated by the

Stalking Horse APA. All entities are authorized and specifically directed to strike all recorded

Interests against the Acquired Assets from their records, official or otherwise.

14. Upon Closing, as and to the extent permitted by applicable law after giving full

force and effect to section 363 of the Bankruptcy Code and this Sale Order, the Stalking Horse

Bidder (and its successors, assigns, agents or representatives) shall not be deemed to be, as a result

of any action or inaction taken in connection with the Sale, (a) a successor to the Debtors, (b) de

facto merged with the Debtors, or (c) a mere continuation of the Debtors. Without limiting the

generality of the foregoing, and except as specifically provided in the Stalking Horse APA, the

Stalking Horse Bidder (and its successors, assigns, agents or representatives) shall not be liable for

any claims against the Debtors or its predecessors or affiliates or their assets, other than as

expressly provided for in the Stalking Horse APA or in this Sale Order.

IV. Assumed Contracts

15. The Debtors are hereby authorized and directed in accordance with sections 105(a),

363, and 365 of the Bankruptcy Code to (a) assume and assign to the Stalking Horse Bidder, in

accordance with the terms of the Stalking Horse APA, the Assumed Contracts free and clear of all

Interests of any kind or nature whatsoever (other than the Permitted Liens, if any, and the Assumed

Liabilities), and (b) execute and deliver to the Stalking Horse Bidder such documents or other

instruments as the Stalking Horse Bidder reasonably deems necessary to assign and transfer the

Assumed Contracts to the Stalking Horse Bidder.

16. With respect to the Assumed Contracts: (a) the Debtors may assume each of the

Assumed Contracts in accordance with section 365 of the Bankruptcy Code; (b) the Debtors may

transfer and assign each Assumed Contract in accordance with sections 363 and 365 of the

30280469.4

Bankruptcy Code, and any provisions in any Assumed Contract that prohibit or condition the

transfer or assignment of such Assumed Contract or allow the party to such Assumed Contract to

terminate, recapture, impose any penalty, condition renewal or extension, or modify any term or

condition upon the assignment of such Assumed Contract, constitute unenforceable anti-

assignment provisions which are void and of no force and effect; and (c) all other requirements

and conditions under sections 363 and 365 of the Bankruptcy Code for the assumption by the

Debtors and assignment to the Stalking Horse Bidder of each Assumed Contract have been

satisfied. Each of the Assumed Contracts shall be transferred and assigned to, and following the

Closing of the Sale and the payment of the applicable Cure Cost, whichever is later, remain in full

force and effect for the benefit of, the Stalking Horse Bidder, notwithstanding any provision in any

such Assumed Contract (including those of the type described in sections 365(c) and (f) of the

Bankruptcy Code) that prohibits, restricts, or conditions such assignment or transfer and, pursuant

to section 365(k) of the Bankruptcy Code, the Debtors shall be relieved from any further liability

with respect to the Assumed Contracts after such assumption and assignment to the Stalking Horse

Bidder.

17. Any defaults or other obligations of the Debtors under the Assumed Contracts

arising or accruing prior to the Closing or required to be paid or performed pursuant to section 365

of the Bankruptcy Code in connection with the assumption and assignment of the Assumed

Contracts, shall be cured by the Stalking Horse Bidder.

18. Upon the Closing, in accordance with sections 363 and 365 of the Bankruptcy

Code, the Stalking Horse Bidder shall be fully and irrevocably vested in all right, title and interest

of each Assumed Contract. Each of the Assumed Contracts shall be assumed and assigned to the

Stalking Horse Bidder free and clear of all liens, claims, or other encumbrances (other than the

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 21 of 30

Assumed Liabilities, the Permitted Liens, if any, or otherwise as set forth in the Stalking Horse

APA).

19. To the extent such party received notice of the Sale, each Contract Counterparty is

deemed to have consented to assumption and assignment and the Stalking Horse Bidder shall be

deemed to have demonstrated adequate assurance of future performance with respect to such

Assumed Contract pursuant to sections 365(b)(1)(C) and 365(f)(2)(B) of the Bankruptcy Code.

20. Upon the Debtors' assignment of the Assumed Contracts to the Stalking Horse

Bidder under the provisions of this Sale Order, any additional orders of this Court, and the payment

of any Cure Costs pursuant to the terms hereof or the Stalking Horse APA, no default shall exist

under any Assumed Contract, and no counterparty to any Assumed Contract shall be permitted (a)

to declare a default by the Stalking Horse Bidder under such Assumed Contract or (b) to otherwise

take action against the Stalking Horse Bidder as a result of any Debtors' financial condition,

bankruptcy, or failure to perform any of its obligations under the relevant Assumed Contract. Each

Contract Counterparty hereby is also forever barred, estopped, and permanently enjoined from (i)

asserting against the Debtors or the Stalking Horse Bidder, or the property of any of them, any

default or claim arising out of any indemnity obligation or warranties for acts or occurrences

arising prior to or existing as of the closing of the Sale, or, against the Stalking Horse Bidder, any

counterclaim, defense, setoff, or any other claim asserted or assertable against the Debtors and (ii)

imposing or charging against the Stalking Horse Bidder or its affiliates any rent accelerations,

assignment fees, increases, or any other fees as a result of the Debtors' assumption and assignment

to the Stalking Horse Bidder of the Assumed Contracts; provided, however, that a Contract

Counterparty shall not be barred from seeking additional amounts on account of any defaults

30280469.4 - 21 -

occurring under the Assumed Contract between the Assumption/Assignment Objection Deadline

(as defined in the Bidding Procedures Order) and the assumption of the Assumed Contract.

21. The Stalking Horse Bidder shall be deemed to be substituted for the Debtors as a

party to the applicable Assumed Contracts, and pursuant to section 365(k) of the Bankruptcy Code,

the Debtors shall be relieved from any further liability under the Assumed Contracts.

22. All Contract Counterparties shall cooperate and expeditiously execute and deliver,

upon the reasonable requests of the Stalking Horse Bidder, and shall not charge the Debtors or the

Stalking Horse Bidder for any instruments, applications, consents, or other documents that may be

required or requested by any public authority or other party or entity to effectuate the applicable

transfers in connection with the sale of the Acquired Assets.

23. Subject to the terms of the Bidding Procedures Order, to the extent that the parties

are unable to consensually resolve the Eversana Cure Objection, unless otherwise agreed by the

parties, the Eversana Cure Objection shall be heard by the Court at such date and time as may be

agreed to by the parties or fixed by the Court.

V. Credit Bid

24. Pursuant to the Stalking Horse APA and the Bidding Procedures Order, and as

authorized by sections 363(b) and 363(k) of the Bankruptcy Code, the Stalking Horse Bidder shall

be entitled to credit bid on the Credit Bid Inventory (as defined in the Bidding Procedures Order)

up to the amount of the Allowed MPG Claim (as defined in the Bidding Procedures Order). At

Closing, the Credit Bid will result in full and final satisfaction of that portion of the Allowed MPG

Claim.

25. Following the Closing of the Sale, the MPG Group shall be deemed to have an

allowed, pre-petition unsecured claim against the Debtors equal to \$1,700,000.00 (the "Remaining"

30280469.4 - 22 -

Stalking Horse Bidder Claim"). The Remaining Stalking Horse Bidder Claim shall not be subject to avoidance, reduction, disallowance, impairment or subordination pursuant to the Bankruptcy Code or applicable non-bankruptcy law and shall be valid and enforceable against any trustee appointed in the Chapter 11 Cases or any successor cases, or upon the dismissal or conversion of

VI. Resolution of Eversana Cure Objection

the Chapter 11 Cases or any successor case.

The Eversana Life Science Services, LLC's Objection to Notice of (I) Potential 26. Assumption and Assignment of Executory Contracts and Unexpired Leases and (II) Cure Amounts [D.I. 108] (the "Eversana Cure Objection") is resolved as follows: Notwithstanding other provisions of this Sale Order, as agreed to by and between the Debtors, the Stalking Horse Bidder, and Eversana Life Science Services, LLC ("Eversana"), the Debtors agree to assume and assign to the Stalking Horse Bidder, on the Closing Date, the "Operational Services Agreement" with an effective date of May 25, 2021 entered into between the Debtors and Eversana. Specifically, the Debtors, the Stalking Horse Bidder, and Eversana agree to proceed with only the Debtors' assumption and assignment to the Stalking Horse Bidder of "Compliance" services outlined in the Operational Services Agreement. The Debtors, the Stalking Horse Bidder, and Eversana agree that the "Channel Management Services" offered under the Operational Services Agreement will not continue after the Closing Date. The "Letter of Engagement" dated February 23, 2021, referenced in the Eversana Cure Objection, which encompasses the "Quality Services" referenced in the Cure Notice, will also be assumed by the Debtors and assigned to the Stalking Horse Bidder. The Operational Services Agreement and Letter of Engagement (and their respective amendments and statements of work) are collectively referred to as the "Eversana Agreements."

30280469.4 - 23 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 24 of 30

27. Eversana has a cure payment owed to it by the Debtors under the Eversana

Agreements of \$30,841.20 (the "Eversana Cure Amount"). Eversana is also entitled to a payment

of \$5,200.00 for the cost associated with the destruction of the 29 pallets currently sitting at the

Eversana facility (the "Destruction Costs"). No later than 10 business days after the Closing Date,

the Stalking Horse Bidder will pay the Eversana Cure Amount and the Destruction Costs less any

amounts Eversana receives from the Debtors in relation thereto. Following the Closing Date and

the payment of the Eversana Cure Amount and the Destruction Costs, whichever is later, without

the need for any further action of the parties or this Court, Eversana shall be deemed to have waived

any and all pre-petition claims against the Debtors and their estates, and Eversana shall not be

treated as a creditor for the purposes of voting and distribution in the Chapter 11 Cases. Eversana

explicitly reserves the right to assert any and all post-petition claims against the Debtors and their

estates on account of services provided by Eversana to the Debtors prior to the Closing Date, and

the Debtors and their estates explicitly reserve any and all rights with respect to any such post-

petition claims.

28. For the avoidance of doubt, upon the Closing Date, the Stalking Horse Bidder will

be Eversana's client under the Eversana Agreements, and the relationship between (a) the Stalking

Horse Bidder and Eversana will continue in accordance with the terms of the Eversana Agreements

and the resolution of the Eversana Cure Objection outlined in this Sale Order and (b) the Debtors

and Eversana shall be deemed consensually terminated without (i) the need for any further action

of the parties or this Court and (ii) any claims resulting from such termination against the Debtors'

and their estates.

30280469.4 - 24 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 25 of 30

VII. Other Provisions

29. Effective upon the Closing Date, the Debtors, on behalf of themselves and their

respective estates, shall forever and irrevocably be deemed to have released, discharged, and

acquitted the Stalking Horse Bidder and each of its former, current, and future officers, employees,

directors, agents, representatives, owners, members, shareholders, partners, financial and other

advisors, investors and consultants, legal advisors, shareholders, managers, consultants, accountants,

attorneys, affiliates, and predecessors and successors in interest from any and all claims, demands,

liabilities, responsibilities, disputes, remedies, causes of action, indebtedness and obligations, rights,

assertions, allegations, actions, suits, controversies, proceedings, losses, damages, injuries,

reasonable attorneys' fees, costs, expenses, or judgments of every type (including, for the avoidance

of doubt, any rights, claims and defenses of the Debtors and their estates with respect to the

Remaining Stalking Horse Bidder Claim), whether known, unknown, asserted, unasserted,

suspected, unsuspected, accrued, unaccrued, fixed, contingent, pending or threatened, including all

legal and equitable theories of recovery, arising under common law, statute or regulation or by

contract, of every nature and description; provided, however, nothing herein shall release any claims

or causes of action that the Debtors may have to enforce the terms of the Stalking Horse APA and

this Sale Order.

30. Effective upon the Closing Date, the Stalking Horse Bidder shall forever and

irrevocably be deemed to have released, discharged, and acquitted the Debtors and their estates and

each of their former, current, and future officers, employees, directors, agents, representatives,

owners, members, shareholders, partners, financial and other advisors, investors and consultants,

legal advisors, shareholders, managers, consultants, accountants, attorneys, affiliates, and

predecessors and successors in interest from any and all claims, demands, liabilities, responsibilities,

30280469.4 - 25 -

disputes, remedies, causes of action, indebtedness and obligations, rights, assertions, allegations, actions, suits, controversies, proceedings, losses, damages, injuries, reasonable attorneys' fees, costs, expenses, or judgments of every type, whether known, unknown, asserted, unasserted, suspected, unsuspected, accrued, unaccrued, fixed, contingent, pending or threatened, including all legal and equitable theories of recovery, arising under common law, statute or regulation or by contract, of every nature and description; provided, however, nothing herein shall release any rights, claims and defenses of the Stalking Horse Bidder with respect to the Remaining Stalking Horse Bidder Claim or any claims or causes of action that the Stalking Horse Bidder may have to enforce the terms of the Stalking Horse APA and this Sale Order.

31. Neither the Stalking Horse Bidder nor its successors and assigns, and their respective current and former affiliates, officers, directors, managers, employees, equity interest holders, agents, representatives, attorneys, and each of their respective successors and assigns, shall have any liability with respect to Excluded Liabilities, and no successor or vicarious liability of any kind or character, whether asserted or unasserted, fixed or contingent, liquidated or unliquidated, with respect to the Debtors or any obligations of the Debtors arising prior to the Closing Date, or in any way related to the operation of the Business or any of the Acquired Assets prior to the Closing Date, except as provided in the Stalking Horse APA. The Stalking Horse Bidder has given substantial consideration under the Stalking Horse Bidder shall constitute valid and valuable consideration for the releases of any potential claims of successor or vicarious liability of the Stalking Horse Bidder, which releases shall be deemed to have been given in favor of the Stalking Horse Bidder by all holders of Interests against or in the Debtors or any of the

30280469.4 - 26 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 27 of 30

Acquired Assets, and which releases shall exclude the Stalking Horse Bidder's obligations under

the Stalking Horse APA and this Sale Order.

32. To the greatest extent available under applicable law, the Stalking Horse Bidder

shall be authorized, as of the Closing Date, to operate under any license, permit, registration and

any other governmental authorization or approval of the Debtors with respect to the Acquired

Assets and the Contracts, and all such licenses, permits, registrations and governmental

authorizations and approvals are deemed to have been, and hereby are transferred to the Stalking

Horse Bidder as of the Closing Date. To the extent provided by section 525 of the Bankruptcy

Code, no governmental unit may revoke or suspend any permit or license related to the operation

of the Acquired Assets sold, transferred or conveyed to the Stalking Horse Bidder on account of

the filing or pendency of the Chapter 11 Cases. For the avoidance of doubt, nothing in this Sale

Order or the Stalking Horse APA authorizes the transfer or assignment of any governmental

(a) license, (b) permit, (c) registration, (d) authorization or (e) approval, or the discontinuation of

any obligation thereunder, without compliance with all applicable legal requirements and

approvals under police or regulatory law.

33. Because the Court has found that the transactions contemplated by the Stalking

Horse APA are undertaken by the Stalking Horse Bidder without collusion and in good faith, as

that term is contemplated by section 363(m) of the Bankruptcy Code, and that the consideration

provided by the Stalking Horse Bidder for the Acquired Assets is fair and reasonable, the reversal

or modification on appeal of the authorization provided herein to consummate the Sale shall not

affect the validity of the Sale (including the assumption and assignment of the Assumed Contracts),

unless such authorization and such Sale are duly stayed pending such appeal. The Stalking Horse

Bidder is a good faith buyer within the meaning of section 363(m) of the Bankruptcy Code and,

30280469.4 - 27 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 28 of 30

as such, is entitled to and is hereby granted the full protections of section 363(m) of the Bankruptcy

Code.

34. The consideration provided by the Stalking Horse Bidder for the Acquired Assets

under the Stalking Horse APA is fair and reasonable under the circumstances, and neither the

Debtors nor the Stalking Horse Bidder have engaged in any action or inaction that would cause or

permit the Sale to be avoided or costs or damages to be imposed under section 363(n) of the

Bankruptcy Code.

Notwithstanding the potential applicability of Bankruptcy Rules 7062, 9014, 35.

6004(h) and 6006(d), this Sale Order shall be effective immediately upon entry and the Debtors

and the Stalking Horse Bidder are authorized to close the Sale, and take any action authorized by

this Sale Order, immediately upon entry of this Sale Order.

36. No bulk sales law or any similar law of any state or other jurisdiction applies in any

way to the Sale.

37. The failure specifically to include any particular provision of the Stalking Horse

APA or the Transaction Documents in this Sale Order shall not diminish or impair the effectiveness

of such provision, it being the intent of the Court that the Stalking Horse APA and the Transaction

Documents be authorized and approved in their entireties.

38. The Stalking Horse APA, the Transaction Documents and any related agreements,

documents or other instruments may be modified, amended or supplemented by the parties thereto

and in accordance with the terms thereof, without further order of the Court, provided that any

such modification, amendment or supplement does not have a material adverse effect on the

Debtors' estates.

- 28 -

30280469.4

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 29 of 30

39. The Stalking Horse Bidder shall have no obligation to proceed with Closing unless

and until all conditions precedent to its obligations to do so, as set forth in the Stalking Horse APA,

have been met, satisfied, or waived in accordance with the terms of the Stalking Horse APA.

40. The provisions of this Sale Order and the Stalking Horse APA, and any actions

taken pursuant hereto or thereto, including under the Transaction Documents, shall survive entry

of any order (a) confirming or consummating any plan of reorganization or liquidation of the

Debtors, (b) converting the Debtors' cases from chapter 11 to chapter 7, (c) dismissing the Chapter

11 Cases, or (d) appointing a chapter 11 trustee or examiner, or the appointment of any other

trustee in any successor cases, and the terms and provisions of the Stalking Horse APA as well as

the rights and interests granted pursuant to this Sale Order and the Stalking Horse APA shall

continue in this or any superseding case and shall be binding upon the Debtors, their affiliates,

estates, and creditors, all holders of equity interests in the Debtors, all holders of any Interests, all

Contract Counterparties, all parties in interest in the Chapter 11 Cases, any trustees or estate

fiduciaries subsequently appointed in the Chapter 11 Cases or upon a conversion to chapter 7 of

the Bankruptcy Code, the Stalking Horse Bidder, and the respective successors and assigns of each

of the foregoing. Nothing contained in any order entered in the Chapter 11 Cases or any related

proceeding subsequent to entry of this Sale Order, nor in any chapter 11 plan confirmed in the

Chapter 11 Cases, shall conflict with or derogate from the provisions of the Stalking Horse APA

or the terms of this Sale Order.

41. The provisions of this Sale Order are non-severable and mutually dependent.

42. All time periods set forth in this Sale Order shall be calculated in accordance with

Bankruptcy Rule 9006(a).

30280469.4 - 29 -

Case 23-10456-MFW Doc 147 Filed 05/25/23 Page 30 of 30

43. To the extent that this Sale Order is inconsistent with the Stalking Horse APA or

any related documents or any prior order or pleading with respect to the Sale Motion in these

Chapter 11 Cases, the terms of this Sale Order shall govern.

44. Effective upon the Closing, the caption in these Chapter 11 Cases shall change to:

> IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

Chapter 11 In re:

PLX PHARMA WINDDOWN CORP., et al.,1

Case No. 23-10456 (MFW)

Debtors.

(Jointly Administered)

¹ The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, are: PLx Pharma Winddown Corp. (5704) and PLx Opco Winddown Corp. (6588). The mailing address for each of the Debtors is 9 Fishers Lane, Suite E, Sparta, NJ 07871.

45. The Court shall retain exclusive jurisdiction over all matters arising from or related

to the interpretation and implementation of this Sale Order, and to interpret, implement and enforce

the terms and provisions of the Stalking Horse APA, the Transaction Documents, all agreements

executed in connection with the Stalking Horse APA to which the Debtors are a party or which

have been assigned by the Debtors to the Stalking Horse Bidder, and to adjudicate any and all

disputes concerning or relating in any way to the Stalking Horse APA, the Sale, or this Sale Order,

including to compel delivery of the Acquired Assets, to protect the Stalking Horse Bidder and its

assets, including the Acquired Assets, against any Interests and successor and transferee liability,

and to enter orders, as appropriate, pursuant to Sections 105, 363, 365, or other applicable

provisions of the Bankruptcy Code to transfer the Acquired Assets and Assumed Contracts to the

Stalking Horse Bidder.

Dated: May 25th, 2023 30280469. Wilmington, Delaware

MARY F. WALRATH - 30 UNITED STATES BANKRUPTCY JUDGE

Exhibit A

Stalking Horse APA (See attached)

30280469.4

Execution Version

ASSET PURCHASE AGREEMENT

dated as of April 12, 2023

by and among

PLX ACQUISITION COMPANY, LLC

as Buyer,

and

PLX PHARMA INC.

and

PLX OPCO INC.,

as Sellers

EAST\202396147.8

Case 23-10456-MFW Doc 147-1 Filed 05/25/23 Page 3 of 63

Table of Contents

		Page:
Article I.	Definitions	1
1.1	Defined Terms.	1
1.2	Interpretation	17
Article II	. Acquired Assets and Assumption Of Liabilities	18
2.1	Assets to be Acquired.	18
2.2	Excluded Assets.	18
2.3	Liabilities to be Assumed by Buyer.	18
2.4	Excluded Liabilities.	18
2.5	Receipt of Misdirected Assets; Liabilities.	18
2.6	Assumption and Assignment of Contracts	19
Article II	I. Closing; Purchase Price	19
3.1	Closing; Transfer of Possession; Certain Deliveries	19
3.2	Consideration.	20
3.3	Deposit.	21
3.4	Allocation of Purchase Price.	22
Article I	V. Representations and Warranties of Sellers	22
4.1	Organization	22
4.2	Authorization of Transactions	22
4.3	Consents and Approvals.	23
4.4	No Conflicts.	23
4.5	Acquired Assets.	23
4.6	Litigation; Orders.	23
4.7	Books and Records.	23
4.8	Compliance with Laws; Permits.	24
4.9	Contracts.	24
4.10	Real Property.	25
4.11	Intellectual Property.	25
4.12	Brokers' Fees and Commissions.	26
4.13	Financial Statements.	26
4.14	Regulatory Matters.	26
4.15	Taxes.	28
4.16	Insurance.	29
4.17	Interested Party Transactions.	29
4 18	Disclosure	29

Case 23-10456-MFW Doc 147-1 Filed 05/25/23 Page 4 of 63

Table of Contents Continued

Article `	V. Representations and Warranties of Buyer	Page: 29
5.1	Organization	
5.2	Due Authorization, Execution and Delivery; Enforceability	30
5.3	Governmental Approvals	
5.4	No Conflicts.	30
5.5	Sufficiency of Funds	31
5.6	Adequate Assurances Regarding Executory Contracts	31
5.7	Exclusive Representations and Warranties.	31
5.8	As Is and Where Is	31
Article `	VI. Covenants of The Parties	31
6.1	Interim Operating Covenants.	31
6.2	Access.	34
6.3	Public Announcements.	35
6.4	Tax Matters.	35
6.5	Commercially Reasonable Efforts.	35
6.6	Further Assurances	36
6.7	Bankruptcy Court Matters	37
6.8	Cure Costs.	39
6.9	Solicitation of Alternative Transactions in Accordance with Bidding Procedures	39
6.10	Notification of Certain Matters.	39
6.11	Confidentiality	40
6.12	Approvals and Consents.	40
6.13	Update of Disclosure Schedules.	41
6.14	Permits.	41
6.15	Bulk Transfer Laws	41
6.16	Buyer Deficiency Claim.	41
6.17	Preservation of Records.	41
6.18	Third-Party Consents.	42
Article `	VII. Conditions to Obligations of The Parties	42
7.1	Conditions Precedent to Obligations of Buyer.	42
7.2	Conditions Precedent to the Obligations of Sellers.	44
7.3	Frustration of Conditions Precedent.	44
Article `	VIII. Termination	44
8.1	Termination of Agreement.	44

Case 23-10456-MFW Doc 147-1 Filed 05/25/23 Page 5 of 63

Table of Contents Continued

Page:	
sequences of Termination46	8.2
scellaneous47	Article E
enses	9.1
gnment	9.2
es in Interest	9.3
ces	9.4
ice of Law48	9.5
re Agreement; Amendments and Waivers	9.6
nterparts; Facsimile and Electronic Signatures	9.7
rability49	9.8
dings	9.9
usive Jurisdiction; Specific Performance	9.10
IVER OF RIGHT TO TRIAL BY JURY50	9.11
ival50	9.12
-Recourse51	9.13
losure Schedules51	9.14
ual Drafting52	9.15
e of Essence52	9.16

ASSET PURCHASE AGREEMENT

This ASSET PURCHASE AGREEMENT, dated as of April 12, 2023 (the "<u>Agreement Date</u>"), is by and among PLx Pharma Inc., a Delaware corporation (the "<u>Company</u>") and PLx Opco Inc., a Delaware corporation and a wholly-owned subsidiary of the Company (together with the Company, "<u>Sellers</u>" and each, a "<u>Seller</u>") and PLx Acquisition Company, LLC, a Delaware limited liability company and wholly-owned subsidiary of Greenwood ("<u>Buyer</u>"). Each Seller and Buyer are referred to herein individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>".

RECITALS

WHEREAS, Sellers are a specialty pharmaceutical company focused on their clinically-validated and patent-protected PLxGuard drug delivery platform, with their lead product being VAZALORE®, an over-the-counter proprietary aspirin product (the "<u>Business</u>");

WHEREAS, Market Performance Group, LLC, a Delaware limited liability company ("MPG") and Greenwood Brands, LLC, a New York limited liability company ("Greenwood"), which are Affiliates of Buyer, provide Sellers with warehousing, distribution, sales, and collections support in respect of VAZALORE® pursuant to various commercial arrangements, including that certain Warehousing and Distribution Services Agreement, dated as of July 1, 2021, that certain Commercial Sales and Support Agreement, dated as of May 1, 2019, and that certain Dashboards and Sales Analyst Support Services Agreement, dated as of May 6, 2020;

WHEREAS, MPG and Greenwood, collectively, assert secured Claims against Sellers in excess of \$7,500,000 in the aggregate as of the date hereof;

WHEREAS, Sellers intend to file voluntary petitions for relief under Chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>") in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>") following the execution of this Agreement (the "<u>Chapter 11 Cases</u>"); and

WHEREAS, Sellers have agreed to sell and assign to Buyer, and Buyer has agreed to serve as a stalking horse purchaser to purchase and assume, the Acquired Assets as of the Closing, subject to higher and better offers Sellers may receive and the terms and conditions set forth in this Agreement and in accordance with sections 105, 363, and 365 and other applicable provisions of the Bankruptcy Code.

NOW, **THEREFORE**, in consideration of the representations, warranties, covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties hereto agree as follows:

ARTICLE I. DEFINITIONS

1.1 <u>Defined Terms</u>. As used herein, the terms below shall have the following respective meanings:

EAST\202396147.8

PATENT REEL: 064913 FRAME: 0402 "Acquired Assets" shall mean all properties, assets, interests and rights of every nature, tangible and intangible of Sellers (real or personal, now or existing or hereafter acquired, whether or not reflected on the books or financial statements of Sellers) relating to the Business, and including the following assets, and <u>provided</u>, that the Excluded Assets shall not be Acquired Assets:

- (a) all of Sellers' rights, title, and interests in and to Inventory;
- (b) all of Sellers' rights, title, and interests in and to accounts receivable, including all trade and non-trade accounts receivable, notes receivable, negotiable instruments, and chattel paper owned or held, together with any unpaid interest of fees accrued thereof or other amounts due with respect thereto;
- (c) all of Sellers' rights, title, and interests in and to the Retailer Escrowed Funds as provided in Section 6.1(b)(iii);
- (d) all Seller Intellectual Property, including (without limitation), those items listed in Section 4.11(a) of the Disclosure Schedules;
 - (e) all equipment and machinery;
 - (f) all Avoidance Actions;
 - (g) all Acquired Claims;
- (h) all third-party warranties, representations, guarantees, refunds, rights of recovery, rights of set-off or counterclaim and rights of recoupment of every kind and nature for the benefit of, or enforceable by, any Seller in each case to the extent arising from or relating to the Business and the Acquired Assets;
- (i) to the extent transferrable under applicable Law, any Tax credits, Tax refunds, Tax deposits, Tax rebates, Tax attributes, and prepaid Tax amounts of the Sellers;
 - (i) the Assumed Contracts;
- (k) all rights of any Seller under nondisclosure or confidentiality, non-compete, or non-solicitation agreements, including with regard to current and former employees and agents of any Seller or with third parties related to the Acquired Assets, the Assumed Liabilities, or the Business (or any portion thereof);
- (l) all insurance policies and all rights of any nature with respect to any such insurance policy, including all claims and proceeds under such insurance policies, and including related letters of credit, cash or other assets that serve as collateral with respect thereto; except for Sellers' directors' and officers' insurance policies, which, for the avoidance of doubt, are defined as "Excluded Assets" hereunder;

- (m) to the extent transferrable, all security deposits and any other deposits held by vendors, trade creditors, or any other party, other than the deposit maintained by Sellers pursuant to the order approving Sellers' 'first-day' utility motion;
- (n) any and all proceeds relating to any and all bonds, letters of credit, guarantees or other security provided by any Seller and any and all insurance, condemnation and similar proceeds;
- (o) all worldwide Governmental Authorizations, to the extent transferable, relating to any Acquired Asset or Assumed Liability, held by Sellers as of the Closing and all rights under, interests in and deposits under the foregoing, and pending applications for any thereof;
- (p) subject to <u>Section 6.17</u>, all Acquired Business Information, in the possession or reasonable control of any Seller and whether in hard or electronic format;
- (q) all goodwill associated with the Business, the Acquired Assets, or the Assumed Liabilities;
 - (r) all prepaid expenses of Sellers relating to any of the Assumed Contracts;
- (s) financial, marketing and business data, pricing and cost information, business and marketing plans, servers, offsite and backup storage, files, correspondence, records, data, plans, reports and recorded knowledge, historical trademark files, prosecution files of any Seller in whatever media retained or stored, including computer programs and disks, in each case used in, held for use in, or relating to the Business or the Acquired Assets, including files in the possession of or under the control of Sellers;
- (t) all of Sellers' rights of publicity and all similar rights, including all commercial merchandising rights; and
- (u) to the extent requested by Buyer and assignable to Buyer under applicable Law, all Regulatory Authorizations and Regulatory Documentation associated with the Product or Seller Intellectual Property (the "<u>Transferred Permits</u>").
- "Acquired Business Information" shall mean all books and records, including financial information, files, ledgers, documentation, instruments, research, reports, papers, data, marketing materials and information, computer codes and sourcing data, supplier lists, cost and pricing information, business plans, and manuals, research and development files, sales or technical literature or similar information that, in each case, relating to the Business or the Acquired Assets, that is in the possession or control of any Seller.
- "Acquired Claims" shall mean all causes of action, lawsuits, judgments, Claims, refunds, rights of recovery, rights of setoff, counterclaims, defenses, demands, remedies, warranty claims, rights to indemnification, contribution, advancement of expenses or reimbursement, or similar rights (whether choate or inchoate, known or unknown, contingent or noncontingent) available to

Sellers or their estates as of the time of Closing against (a) Buyer or any of its Affiliates (other than Claims pursuant to this Agreement or arising out of the Transactions contemplated hereby), (b) any person who at any time at or prior to Closing served or may serve as a director, officer, manager, employee or advisor of any Seller and any shareholder or Related Party of any Seller, (c) any customer, supplier, manufacturer, distributor, broker, or vendor of any Seller or any other Person with whom any Seller has an ordinary course commercial relationship, and (d) any other third party.

"Affiliate" shall, with respect to any Person, mean any other Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, the Person specified, where "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management policies of a Person, through ownership of voting securities or rights, by contract, as trustee, executor or otherwise.

"Agreement" shall mean this Asset Purchase Agreement, together with the exhibits and the Disclosure Schedules, in each case as amended, restated, supplemented or otherwise modified from time to time.

- "Agreement Date" shall have the meaning specified in the Preamble.
- "Allocation" shall have the meaning specified in <u>Section 3.4</u>.
- "<u>Alternative Transaction</u>" shall mean (a) the approval by the Bankruptcy Court of a sale or sales of any portion of the Acquired Assets to a Person other than Buyer, or (b) the filing of a chapter 11 plan that does not contemplate the sale of the Acquired Assets to Buyer in accordance with the terms hereof.
 - "Assumed Contracts" shall have the meaning specified in Section 2.6(a).
- "Assumed Liabilities" shall mean the Liabilities, including Cure Costs to the extent applicable, of Sellers that Buyer may agree to assume, in its sole discretion, pursuant to the terms of this Agreement and the Sale Order, and as set forth in Section 1.1(a) of the Disclosure Schedules.
 - "Auction" shall have the meaning set forth in the Bidding Procedures Order.
- "Avoidance Actions" shall mean any and all avoidance claims or causes of action available to Sellers under Chapter 5 of the Bankruptcy Code (including sections 544, 545, 547, 548, 549, 550 and 553) or any similar actions under any other applicable Law (whether or not asserted as of the Closing Date).
- "<u>Back-Up Bidder</u>" shall have the meaning specified in <u>Section Error! Reference source</u> not found.
 - "Bankruptcy Code" shall have the meaning specified in the Recitals.
 - "Bankruptcy Court" shall have the meaning specified in the Recitals.

- "Bidding Procedures" shall mean bidding procedures approved by the Bankruptcy Court pursuant to the Bidding Procedures Order, which shall be in form and substance acceptable to Buyer.
- "<u>Bidding Procedures Motion</u>" shall mean the motion of Sellers seeking entry of the Bidding Procedures Order and Sale Order, which shall be in form and substance acceptable to Buyer.
- "<u>Bidding Procedures Order</u>" shall mean an Order of the Bankruptcy Court approving the Bidding Procedures, which shall be in form and substance acceptable to Buyer.
 - "Bid Protections" shall mean (i) the Break-Up Fee, and (ii) Expense Reimbursement.
- "<u>Bill of Sale and Assignment and Assumption Agreement</u>" shall mean a bill of sale and assignment and assumption agreement to be entered into by Sellers and Buyer concurrently with the Closing, substantially in the form of <u>Exhibit A</u>.
 - "Break-Up Fee" shall be an amount equal to \$100,000.
 - "Business" shall have the meaning specified in the Recitals.
- "Business Day" shall mean any day other than a Saturday, Sunday or a legal holiday on which banking institutions in the State of New York are not required to open.
 - "Buyer" shall have the meaning specified in the Preamble.
 - "Cash Consideration" shall have the meaning specified in Section 3.2(a).
 - "Chapter 11 Cases" shall have the meaning specified in the Recitals.
- "Claim" shall mean all actions, claims, counterclaims, suits, proceedings, rights of action, causes of action, Liabilities, losses, damages, remedies, penalties, judgments, settlements, costs, expenses, fines, disbursements, demands, reasonable costs, fees and expenses of counsel, including in respect of investigation, interest, demands and actions of any nature or any kind whatsoever, known or unknown, disclosed or undisclosed, accrued or unaccrued, matured or unmatured, choate or inchoate, legal or equitable, and arising in tort, contract or otherwise, including any "claim" as defined in the Bankruptcy Code.
 - "Closing" shall have the meaning specified in Section 3.1(a).
 - "Closing Date" shall have the meaning specified in Section 3.1(a).
 - "Code" shall mean the Internal Revenue Code of 1986, as amended.
 - "Company" shall have the meaning specified in the Recitals.
- "<u>Confidentiality Agreement</u>" shall mean that certain Confidentiality Agreement, dated January 10, 2018, between the Company and Buyer.

"Contract" shall mean any contract, agreement, franchise agreement, indenture, note, bond, loan, instrument, conditional sales contract, purchase order, mortgage, license, franchise, insurance policy, letter of credit, commitment or other binding arrangement or commitment, whether or not in written form, that is binding upon a Person or any of its property.

"Contracting Parties" shall have the meaning specified in Section 9.13.

"Copyrights" shall have the meaning specified in the definition of "Intellectual Property"

"Credit Bid Consideration" shall have the meaning specified in Section 3.2(a).

"<u>Cure Costs</u>" shall mean all amounts required to be paid pursuant to section 365(b) of the Bankruptcy Code in connection with the assumption and assignment of the Assumed Contracts, or otherwise to effectuate, pursuant to the Bankruptcy Code, the assumption and assignment of the Assumed Contracts.

"Deposit" shall have the meaning specified in Section 3.3(a).

"<u>Disclosure Schedules</u>" shall mean the disclosure schedules, delivered by Sellers and Buyer concurrently with the execution and delivery of this Agreement, as amended from time to time in accordance with and subject to the terms hereof.

"Escrow Account" shall mean the account established by the Escrow Agent to hold the Deposit.

"Escrow Agent" shall mean Young Conaway Stargatt & Taylor, LLP.

"<u>Escrow Agreement</u>" shall mean the escrow agreement by and among Escrow Agent, Buyer, and Sellers.

"<u>Excluded Assets</u>" shall mean all assets of the Sellers that are not Acquired Assets as specifically set forth herein, except for:

- (a) all cash on hand, and bank accounts, of the Sellers as of the Closing;
- (b) all assets and rights of every nature related to benefit plans of Sellers;
- (c) each Excluded Contract;
- (d) all furniture, office supplies and other tangible personal property located at Sellers' leased real property;
- (e) capital stock of Sellers or any securities convertible into, exchangeable, or exercisable for shares of capital stock or other equity interest of each Seller;
- (f) all of Sellers' certificates of incorporation and other organizational documents, arrangements with registered agents relating to foreign qualifications, taxpayer and

other identification numbers, seals, stock transfer books, stock certificates, and other documents primarily relating to the organization, maintenance and existence of any Seller;

- (g) any (i) confidential personnel and medical records pertaining to any employees of Sellers, or (ii) other records, in either case, that Sellers are required by Law to retain, including Tax Returns, taxpayer and other identification numbers, financial statements and corporate or other entity filings; provided that, to the extent permitted by applicable Law, Buyer shall have the right to obtain copies of any portions of such retained Records and documents to the extent that such portions relate to the Business or any Acquired Asset or are otherwise necessary for Buyer to comply with applicable Law;
 - (h) any Retained Privileged Materials;
- (i) Sellers' directors and officers' insurance policies, and all rights and benefits of any nature of the Sellers with respect thereto, including all insurance recoveries thereunder, credits, refunds and rights to assert claims with respect to any such insurance recoveries thereunder;
- (j) Sellers' claims or other rights under this Agreement, including the right to be paid the Purchase Price hereunder at the Closing in accordance with the terms hereof, or Sellers' rights under any agreement, certificate, instrument, or other document executed and delivered between any Seller and Buyer in connection with the transactions contemplated hereby entered into on or after the date hereof; and
- (k) any other asset of Sellers listed in <u>Section 1.1(b)</u> of the Disclosure Schedules.

"Excluded Contracts" shall mean all Contracts other than Assumed Contracts.

"<u>Excluded Liabilities</u>" shall mean any Liabilities of Sellers and their Affiliates that are not Assumed Liabilities, including:

- (a) Excluded Taxes;
- (b) all Liabilities relating to employees of Sellers, including, but not limited to:
- (i) all payroll and benefits, including accrued and unused vacation hours, sick time, or other paid time off, deferred compensation, retention, incentive, or severance programs, or any other compensation arrangement;
- (ii) all Liabilities, if any, under the WARN Act related to the Transactions;
- (iii) all COBRA obligations and related Liabilities with respect to all covered current and former employees of a Seller and their beneficiaries;
 - (c) any Cure Costs in excess of the Cure Cost Cap; and

(d) all other Liabilities set may be forth in <u>Section 1.1(c)</u> of the Disclosure Schedules.

"Excluded Taxes" shall mean (i) all Taxes of any Seller or any Affiliate of any Seller or for which any Seller or any Affiliate of any Seller is liable, for any taxable period, (ii) all Taxes relating to the Acquired Assets, the Business, or Assumed Liabilities arising from or related to a Pre-Closing Tax Period; (iii) half of all Transfer Taxes related to the Transactions, subject to Section 6.4; (iv) Taxes that arise out of failure of any Seller or any Affiliate of any Seller to comply with the requirements and provisions of any bulk sales, bulk transfer, or similar laws of any jurisdiction arising from or related to a Pre-Closing Tax Period; and (v) other Taxes of any Seller or any Affiliate of any Seller of any kind or description (including any Liability for Taxes of any Seller or any Affiliate of any Seller) that becomes a Liability of Buyer under any common law doctrine of de facto merger or transferee or successor liability or otherwise by operation of Contract or Law.

"Expense Reimbursement" shall mean all reasonable and documented out-of-pocket costs, fees, and expenses incurred by Buyer since March 1, 2023, in connection with the Transactions (including, without limitation, attorney fees and any other amounts incurred to accountants and other professionals) up to an aggregate amount of \$400,000.

"FD&C Act" shall have the meaning specified in the definition of "Health Law".

"FDA" shall mean the United States Food and Drug Administration, or any similar Governmental Entity.

"Federal Health Care Program" shall mean any "federal health care program" as such term is defined in 42 U.S.C. §1320a-7b(f), including Medicare (including Medicare Part D and Medicare Advantage), Medicaid and any state health plan adopted pursuant to Title XIX of the Social Security Act, the Children's Health Insurance Program (CHIP), CHAMPUS/ TRICARE and similar or successor programs that are funded, in whole or in part, by or for the benefit of the government of the United States of America.

"First Deposit Payment" shall have the meaning specified in Section 3.3(a).

"Fraud" shall mean an actual and intentional misrepresentation of material facts with respect to (i) the making of any representation or warranty of Sellers set forth in Article 4 or in any other Transaction Document, (ii) the making of the covenants or agreements by Sellers in this Agreement or in any other Transaction Document, or (iii) the certifications of Sellers or Buyer, respectively, set forth in the certificates delivered by Sellers and Buyer pursuant to Section 7.1(d) and Section 7.2(d), respectively, in each case which satisfies all of the elements of common law fraud under applicable Law.

"GAAP" shall mean generally accepted accounting principles as in effect from time to time in the United States.

"Governmental Authorization" shall mean all filings with any Governmental Entity, consents, approvals, authorizations, or notices (to the extent required from a Governmental Entity), licenses, franchises, permits, concessions, exemptions, orders, certificates, registrations, reregistrations, applications, declarations and filings pertaining to the aforesaid issued, granted, given or otherwise made available by or under the authority of any Governmental Entity or pursuant to any applicable Laws, legislation, directives, measures, treaties, conventions and other agreements between states, or between states and supranational bodies, and rules of common or civil law, in each case, having the force of law and having effect in any jurisdiction.

"Governmental Entity" or "Governmental Entities" shall mean any federal, state, provincial, local, municipal, foreign, multinational, international or other (a) government, (b) governmental or quasi-governmental authority of any nature (including any governmental agency, branch, department, official, or entity and any court, arbiter, or other tribunal), or (c) body exercising, or entitled to exercise, any administrative, executive, judicial, legislative, police, regulatory, or taxing authority or power of any nature, including any arbitration tribunal or stock exchange.

"Greenwood" has the meaning specified in the Recitals.

"Health Law" or "Health Laws" shall mean all Laws applicable to the Sellers and the Business, including those of which the purpose is to ensure the safety, efficacy and quality of medicines, biological products or pharmaceuticals by regulating the research, development, manufacturing and distribution of these products, the Federal Food, Drug, and Cosmetic Act (21 U.S.C. § 301 et seq) as amended from time to time ("FD&C Act") together with all Laws promulgated thereunder or enforced by the FDA, the Anti-Kickback Statute (42 U.S.C. section 1320a-7b(b)), the Civil Monetary Penalties Law (42 U.S.C. § 1320a-7a), the Physician Payment Sunshine Act (42 U.S.C. § 1320a-7h), the Civil False Claims Act (31 U.S.C. section 3729 et seq.), the criminal False Claims Law (42 U.S.C. § 1320a-7b(a)), all criminal laws relating to health care fraud and abuse, including but not limited to 18 U.S.C. sections 286 and 287, and the health care fraud criminal provisions under HIPAA (42 U.S.C. section 1320d et seq.), the exclusion laws (42 U.S.C. § 1320a-7), HIPAA and similar state and foreign privacy and data security laws such as the European Union General Data Protection Regulation, Medicare (Title XVIII of the Social Security Act), Medicaid (Title XIX of the Social Security Act), and any other federal, state or non-U.S. Law that regulates the design, development, testing, studying, manufacturing, processing, storing, importing or exporting, licensing, labeling or packaging, advertising, distributing, selling, pricing, or marketing of pharmaceutical products, or that is related to remuneration (including ownership) to or by physicians or other health care providers (including kickbacks) or the disclosure or reporting of the same, licensure or accreditation or any other aspect of providing health care products or services.

"HIPAA" shall mean the Health Insurance Portability and Accountability Act of 1996, as amended and supplemented by the Health Information Technology for Clinical Health Act of the American Recovery and Reinvestment Act of 2009, and as otherwise may be amended from time to time by Congress and/or rulemaking authority of the Secretary of the Department of Health and Human Services, and all rules and regulations promulgated thereunder, including the Privacy Standards (45 C.F.R. Parts 160 and 164), the Electronic Transactions Standards (45 C.F.R. Parts

160 and 162), the Security Standards (45 C.F.R. Parts 160, 162 and 164), and the Breach Notification Rule (45 C.F.R. Parts 160 and 164 Parts A and D).

"Intellectual Property" shall mean any and all intellectual property or proprietary rights in any jurisdiction throughout the world, including (without limitation) all legal rights, title or interest in the following arising under Law: (i) all Patents; (ii) all copyrights, copyright registrations and copyright applications, copyrightable works, works of authorship, mask works, mask work registrations and mask work applications and all other corresponding rights (collectively, "Copyrights"); (iii) all advertising material, social media accounts, social media identifiers (such as a Twitter® handle); (iv) trademarks (whether registered, unregistered or applied-for), service marks, trade names, service names, corporate entity names, brand names, all trade dress rights, logos and corporate names and general intangibles of a like nature, together with the goodwill associated with any of the foregoing, and all applications, registrations and renewals thereof, including any intent to use applications, supplemental registrations and any renewals or extensions, all other indicia of commercial source or origin and all goodwill associated with any of the foregoing (collectively, "Trademarks"); (v) websites, internet addresses and domain names; (vi) all Know-How, research and development information, financial, marketing and business data, business and marketing plans, correspondence, records, and other documentation, and other proprietary information of every kind; (vii) all licenses and permits to the extent transferable; (viii) all rights to market any products in any jurisdiction; (ix) product or clinical data; (x) software (including source code, executable code, systems, tools, data, databases, firmware, and related documentation) (collectively, "Software"); (xi) to the extent not provided in the above, all technical and manufacturing documentation related to the Product and all Regulatory Documentation; (xii) all rights and powers to assert, defend and recover title to any of the foregoing; (xiii) all rights to assert, defend, sue, and recover damages for any past, present and future infringement, misuse, misappropriation, impairment, unauthorized use or other violation of any rights in or to any of the foregoing; (xiv) all other intellectual property rights irrespective of not being registered or applied for registration, in each instance together with all associated goodwill in any jurisdiction in the world; and (xv) copies and tangible embodiments or descriptions of any of the foregoing (in whatever form or medium).

"Intellectual Property Assignment Agreements" shall have the meaning specified in Section 3.1(b)(iii).

"Interim Period" shall have the meaning specified in Section 6.1.

"Inventory" shall mean all raw materials, active pharmaceutical ingredients, excipients, work-in-process, semi-finished and finished goods and Products, supplies (including clinical drug supplies), samples, components, packaging materials, and other inventories owned by Sellers.

"IOI" shall have the meaning specified in Section 7.1(f).

"Know-How" shall mean any and all data, results, and information of any type whatsoever, whether tangible or intangible and regardless of the form or medium, including any know-how, trade secrets, expertise, knowledge, practices, techniques, concepts, methodologies, methods, processes, protocols, designs, ideas, inventions (whether or not patentable or reduced to practice),

improvements, industrial designs and models, discoveries, developments, unpublished patent applications, specifications, formulations, formulae, materials or compositions of matter of any type or kind (patentable or otherwise), assays, screens, software, algorithms, models, data, databases, database rights, chemistry, manufacturing and control (CMC) information and data, lab notebooks, Patent data, stability, technology, test data and results (including pharmacological, biological, chemical, biochemical, toxicological, pre-clinical and clinical test data), analytical and quality control data, results or descriptions, studies and procedures, development, manufacturing and distribution costs, information contained in submissions to and information from regulatory authorities, and marketing and other reports.

"Knowledge" shall mean, with respect to any Seller, the actual knowledge of Natasha Giordano and Rita O'Connor, after reasonable inquiry.

"<u>Law</u>" shall mean any federal, state, provincial, local, foreign, international or multinational constitution, statute, law, ordinance, regulation, rule, code, Order, guidance, policy, guideline, licensing requirement, pronouncement, principle of common law, or decree enacted, promulgated, issued, enforced or entered by any Governmental Entity, or court of competent jurisdiction, or other requirement or rule of law.

"<u>Liabilities</u>" shall mean, as to any Person, all debts, adverse Claims, liabilities, commitments, responsibilities, and obligations of any kind or nature whatsoever, direct or indirect, absolute or contingent, whether accrued or unaccrued, vested or otherwise, liquidated or unliquidated, whether known or unknown, and whether or not actually reflected, or required to be reflected, in such Person's balance sheet or other books and records.

"<u>Lien</u>" shall have the meaning specified in section 101(37) of the Bankruptcy Code and shall include any pledge, option, charge, lien, license, debentures, trust deeds, hypothecation, easement, security interest, right-of-way, encroachment, mortgage, deed of trust, defect of title, restriction on transferability, restriction on use or other encumbrance, in each case whether imposed by agreement, law, equity or otherwise.

"Material Contracts" shall have the meaning specified in Section 4.9(a).

"Medicaid" shall mean, collectively, the health care assistance program established by Title XIX of the Social Security Act (and any statutes succeeding thereto, and all Laws pertaining to such program, including (i) all federal statutes affecting such program; (ii) all state statutes and plans for medical assistance enacted in connection with such program and federal rules and regulations promulgated in connection with such program; and (iii) all applicable provisions of all rules, regulations, manuals, orders and administrative guidance, reimbursement, and requirements of all Governmental Entities promulgated in connection with such program, in each case as the same may be amended, restated, supplemented or otherwise modified from time to time.

"Medicare" shall mean, collectively, the health insurance program for the aged and disabled established by Title XVIII of the Social Security Act and any statutes succeeding thereto, and all Laws, manuals, or requirements pertaining to such program including (i) all federal statutes (whether set forth in Title XVIII of the Social Security Act or elsewhere) affecting such program;

and (ii) all applicable provisions of all rules, regulations, manuals, orders, administrative guidance, reimbursement and requirements of all Governmental Entities promulgated in connection with such program, in each case, as the same may be amended, restated, supplemented or otherwise modified from time to time.

"Milestones" shall have the meaning specified in Section 6.7(d).

"MPG" shall have the meaning specified in the Recitals.

"Non-Party Affiliates" shall have the meaning specified in Section 9.13.

"Notices" shall have the meaning specified in Section 9.4.

"Order" shall mean any judgment, order, injunction, writ, ruling, decree, stipulation, determination, decision, verdict, or award of any Governmental Entity.

"Ordinary Course of Business" shall mean for any Person, an action taken in the ordinary course of business of such Person, consistent with past practices prior to the Petition Date.

"Outside Date" shall mean the date that is the earlier of (a) May 31, 2023, and (b) two (2) Business Days after the entry of the Sale Order; provided, however, that in the event of an objection to entry of the Sale Order or if the Buyer does not waive the fourteen-day period required for the Sale Order to become final, the Outside Date shall be extended to June 13, 2023.

"Party" or "Parties" shall have the meaning specified in the Preamble.

"Patents" shall mean all issued patents, pending patent applications, including utility and design patents, statutory invention registrations and patent disclosures, and all reissues, continuations, continuations-in-part, divisionals, substitutions, extensions, renewals, supplementary protection certificates, reexaminations, renewals extensions and all foreign equivalents of any of the foregoing.

"<u>Permits</u>" shall mean permits, licenses, registrations, consents, approvals, certificates, variances, accounts, clearances and other authorizations issued by any Governmental Entity relating to the Acquired Assets or the Business, to the extent transferrable.

"<u>Permitted Liens</u>" shall mean, with respect to the Acquired Assets, statutory liens of landlords, carriers, warehousemen, mechanics, and materialmen incurred in the Ordinary Course of Business for sums not yet due.

"<u>Person</u>" shall mean an individual, a partnership, a joint venture, a corporation, a business trust, a limited liability company, a trust, an unincorporated organization, a joint stock company, a labor union, an estate, a Governmental Entity or any other entity.

"<u>Personally Identifiable Information</u>" shall mean all personally identifiable information held by, or under the control of Sellers, or used by Sellers.

"<u>Petition Date</u>" shall mean the date that the Sellers file their voluntary petitions for relief under Chapter 11 of the Bankruptcy Code in the Bankruptcy Court.

"Private Program" shall mean those private, non-Federal Health Care Programs, health care plans or health care reimbursement, including private insurance, health maintenance organizations, accountable care organizations, clinically integrated networks, employer-sponsored programs under ERISA, and physician hospital organizations, under which hospitals, healthcare providers, and any healthcare facilities, centers, and units, directly or indirectly, receive payment for the provision of services or goods to beneficiaries of the applicable program.

"Product" shall mean VAZALORE® 325 mg and VAZALORE® 81 mg liquid-filled aspirin capsules. For the avoidance of doubt, the use of the term "Product" in this Agreement does not preclude that more than one Product may be developed, manufactured and/or commercialized utilizing Sellers' PLxGuard drug delivery platform.

"Post-Closing Covenant" shall have the meaning specified in Section 9.12.

"<u>Pre-Closing Tax Period</u>" shall mean any Tax period (or portion thereof) ending on or before the Closing Date.

"Proceeding" shall mean any action, arbitration, audit, known investigation (including a notice of preliminary investigation or formal investigation), notice of violation, hearing, litigation or suit (whether civil, criminal or administrative), other than the Chapter 11 Cases, commenced, brought, conducted or heard by or before any Governmental Entity, including but not limited to any and all such actions related to restitution or remission in criminal proceedings and civil forfeiture and confiscation proceedings under the Law of any jurisdiction.

"Purchase Price" shall have the meaning specified in Section 3.2(a).

"Qualified Bid" or "Qualified Bids" shall have the meaning set forth in the Bidding Procedures Order, which, for the avoidance of doubt, will require that for a third party bid to be deemed a "Qualified Bid", it must include, among other things, (a) a cash component of its purchase price that is sufficient in amount to pay upon the closing of such Alternative Transaction, directly to Buyer as the Stalking Horse (i) the Break-Up Fee, (ii) the Expense Reimbursement, and (iii) the full amount of the Transaction Value set forth herein, and (b) be in an initial amount in cash of at least One Hundred Thousand Dollars (\$100,000) in excess of the aggregate sum of (x) the Transaction Value <u>plus</u> (y) the Expense Reimbursement <u>plus</u> (z) the Break-Up Fee.

"Regulatory Authority" shall mean any national or supranational Governmental Entity, including the FDA, with responsibility for granting any license, registrations, approvals or other Permit (whether transferable or not) with respect to the Products.

"Regulatory Authorizations" shall mean any approvals, clearances, authorizations, exemptions, variances, registrations, listings, certifications, licenses and permits granted by any Regulatory Authority.

"Regulatory Documentation" shall mean all Governmental Authorizations, regulatory, scientific and technical data, documents, and information, and any books, records, files and documents owned, maintained or in the possession or control of Sellers as of the Closing with respect to the development, pre-clinical and clinical testing, manufacturing, packaging, promoting, marketing, selling, distributing and/or otherwise commercializing of the Compound or any of the Acquired Assets, including (i) copies of the applications, approvals, dossiers and submissions to and correspondence to and from the Governmental Entities responsible for the grant of a Regulatory Submission (including minutes and contact reports relating to any communications with any Governmental Entities); (ii) the list of the composition or compositions thereof; (iii) all scientific information underlying contemplated Regulatory Submission(s) referencing the Acquired Assets, including pre-clinical, clinical and other reports and publications related to characterizing the safety and efficacy of the Acquired Assets; (iv) information concerning the Acquired Assets prepared for health care professionals; (v) current marketing, advertising, training materials and other marketing materials, including market access and patient support materials (and any copyrights, know-how, trade secrets or other intellectual property rights in such materials), if any; and (vi) pharmacovigilance databases and any other records relating to safety reporting or data contained or referenced in or supporting any of the foregoing, in any format.

"Regulatory Submission" shall mean any pending or approved submission to FDA or any other Governmental Entity, held by Sellers or any third party authorizing the sourcing, receipt, storage, processing, manufacturing, packaging, labeling, holding or commercialization of a Product, including without limitation, a New Drug Application ("NDA"), Abbreviated New Drug Application ("ANDA"), Investigational New Drug Application ("IND"), or equivalent submissions to any foreign Governmental Entity.

"Related Party" shall mean, with respect to any Person, such Person's Affiliates, successors and assigns and the partners, shareholders, members, investors and potential investors, parents, predecessors, subsidiaries, controlling persons, current and former directors, current and former officers, employees, agents, trustees, administrators, managers, advisors, attorneys and representatives of such Person and of such Person's Affiliates, successors and assigns.

"Representative" shall mean, with respect to any Person, such Person's officers, directors, managers, employees, agents, representatives and financing sources (including any investment banker, financial advisor, accountant, legal counsel, consultant, other advisor, agent, representative or expert retained by or acting on behalf of such Person or its subsidiaries or Affiliates).

"Retailers" shall mean all retailers of Products, including but not limited to, Target, Walmart, Walgreens, CVS, Rite Aid, and certain grocery store retailers.

"Retailer Escrowed Funds" shall have the meaning specified in Section 6.1(b)(iii).

"<u>Retained Privileged Materials</u>" all books and records that are subject to any attorney-client privilege and the transfer of which to Buyer would result in the waiver of any such privilege.

"Sale Order" shall mean an Order of the Bankruptcy Court acceptable to Buyer in all respects pursuant to, *inter alia*, sections 105, 363 and 365 of the Bankruptcy Code authorizing and

approving, inter alia, the sale of the Acquired Assets to Buyer on the terms and conditions set forth herein, which shall, among other things: (a) approve the consummation of the Transactions pursuant to the Transaction Documents; (b) approve the assumption and assignment to Buyer of the Assumed Contracts; (c) find that the sale by Sellers to Buyer of the Acquired Assets is free and clear of all Claims and Liens (except Permitted Liens); (d) find that Buyer is a "good faith" buyer within the meaning of section 363(m) of the Bankruptcy Code and grant Buyer the protections of section 363(m) of the Bankruptcy Code; and (e) approve customary releases for the benefit of Buyer and its directors, officers, employees, affiliates, members, agents, attorneys, and other Representatives.

"SEC" shall have the meaning specified in the preamble to Article IV.

"Second Deposit Payment" shall have the meaning specified in Section 3.3(a).

"Seller" or "Sellers" shall have the meaning specified in the Preamble.

"Seller Intellectual Property" shall mean all Seller Owned Intellectual Property, Seller IP Licenses, and all other Intellectual Property, including Intellectual Property licensed by Sellers, used or held for use in the Business or material to or otherwise necessary for the operation of the Business, including for the avoidance of doubt (without limitation), all rights to market the Product in any jurisdiction, Product data, Product packaging and advertising materials, and all Software or similar data related to Sellers' Amazon Seller Central account to the extent transferrable under applicable Law.

"Seller IP Licenses" shall have the meaning specified in Section 4.11(c).

"Seller IP Registrations" shall have the meaning specified in Section 4.11(a).

"Seller Fundamental Representations" shall have the meaning specified in Section 7.1(a).

"Seller Material Adverse Effect" shall mean any event, change, development or occurrence that, individually or together with any other event, change, development or occurrence, is material and (a) adverse to the Business, condition (financial or otherwise), assets, or results of operations of Sellers; or (b) would adversely affect the ability of Sellers to consummate the Transaction contemplated in this Agreement on a timely basis. The term "Material Adverse Effect" shall not include any event, occurrence, fact, condition or change, directly or indirectly, arising out of or attributable to: (i) general economic or political conditions; (ii) any changes in financial or securities markets in general; (iii) acts of war (whether or not declared), armed hostilities or terrorism, or the escalation or worsening thereof; (iv) any changes in applicable Laws or accounting rules, so long as, in each case of clauses (i) through (iv), the Business is not disproportionately affected by such effect, event or condition as compared to industry peers.

"<u>Seller Owned Intellectual Property</u>" shall mean all Intellectual Property owned or purported to be owned by, or exclusively licensed by, Sellers and used in or held for use in the Business or material to or otherwise necessary for the operation of the Business.

"Stalking Horse" shall have the meaning specified in Section 6.7(d)(v).

"Successful Bidder" shall have the meaning set forth in the Bidding Procedures Order.

"Tax" or "Taxes" shall mean any and all taxes, assessments, levies, duties or other governmental charge imposed by any Governmental Entity, including any income, alternative or add-on minimum, accumulated earnings, franchise, capital stock, unclaimed property or escheatment, environmental, profits, windfall profits, gross receipts, sales, use, highway use, fuel, vehicle registration, value added, transfer, registration, stamp, premium, excise, customs duties, severance, personal property, ad valorem, occupancy, license, occupation, employment, payroll, social security, disability, unemployment, withholding, corporation, inheritance, value added, stamp duty reserve, estimated or other tax, assessment, levy, duty (including duties of customs and excise) or other governmental charge of any kind whatsoever, including any payments in lieu of taxes or other similar payments, chargeable by any Tax Authority together with all penalties, interest and additions thereto, whether disputed or not.

"<u>Tax Authority</u>" shall mean any taxing or other authority (whether within or outside the U.S.) competent to impose Tax.

"<u>Tax Return</u>" shall mean any and all returns, declarations, reports, documents, Claims for refund, or information returns, statements or filings which are supplied or required to be supplied to any Tax Authority or any other Person, including any schedule or attachment thereto, and including any amendments thereof.

"Trademarks" shall have the meaning specified in the definition of "Intellectual Property".

"<u>Third Party Payor</u>" shall mean any Federal Health Care Program, Private Program, or any other Person that maintains Third Party Payor Programs.

"<u>Third Party Payor Programs</u>" shall mean all payment or reimbursement programs sponsored or maintained by any Third Party Payor.

"<u>Transaction Documents</u>" shall mean this Agreement, the Bill of Sale and Assignment and Assumption Agreement, the Intellectual Property Assignment Agreements, the Escrow Agreement, and each other agreement, document or instrument executed and delivered by the Parties in connection with the consummation of the Transactions.

"Transaction Value" shall have the meaning specified in Section 3.2(a).

"<u>Transactions</u>" shall mean the transactions contemplated by this Agreement, including the purchase and sale of the Acquired Assets as provided for in this Agreement.

"Transfer Tax" or "Transfer Taxes" shall mean any sales, use, transfer, conveyance, documentary transfer, stamp, recording or other similar Tax imposed upon the sale, transfer or assignment of property or any interest therein or the recording thereof, and any penalty, addition to Tax or interest with respect thereto, but such term shall not include any Tax on, based upon or

measured by, the net income, gains or profits from such sale, transfer or assignment of the property or any interest therein.

"Transferred Permits" shall have the meaning specified in the definition of "Acquired Assets".

- "<u>U.S. Person</u>" shall mean any Person that is a "United States person" as defined in section 7701(a)(30) of the Code.
- 1.2 <u>Interpretation</u>. Whenever the words "include," "includes" or "including" are used in this Agreement they shall be deemed to be followed by the words "without limitation."
- (b) Words denoting any gender shall include all genders. Where a word or phrase is defined herein, each of its other grammatical forms shall have a corresponding meaning.
- (c) A reference to any Party to this Agreement or any other agreement or document shall include such Party's successors and permitted assigns. Solely with regard to the Credit Bid Consideration, references to "Buyer" shall be inclusive of MPG and Greenwood, as applicable.
- (d) A reference to any legislation or to any provision of any legislation shall include any modification or re-enactment thereof, any legislative provision substituted therefor, and all regulations and statutory instruments issued thereunder or pursuant thereto.
- (e) All references to "\$" and dollars shall be deemed to refer to United States currency.
- (f) All references to any financial or accounting terms shall be defined in accordance with GAAP.
- (g) The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section, Disclosure Schedule and exhibit references are to this Agreement unless otherwise specified. All article, section, paragraph, schedule and exhibit references used in this Agreement are to articles, sections and paragraphs of, and schedules and exhibits to, this Agreement unless otherwise specified.
- (h) The meanings given to terms defined herein shall be equally applicable to both singular and plural forms of such terms.
- (i) When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is not a Business Day, the period in question shall end on the next succeeding Business Day. All references herein to time are references to New York City time, unless otherwise specified herein.

- (j) If a word or phrase is defined, its other grammatical forms have a corresponding meaning.
- (k) A reference to any agreement or document (including a reference to this Agreement) is to the agreement or document as amended or supplemented, except to the extent prohibited by this Agreement or that other agreement or document.
- (l) Exhibits and Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.

ARTICLE II. ACQUIRED ASSETS AND ASSUMPTION OF LIABILITIES

- 2.1 <u>Assets to be Acquired</u>. Subject to entry of the Sale Order, and the terms and conditions of this Agreement and the Sale Order, at the Closing, Sellers shall sell, assign, transfer and deliver to Buyer, and Buyer shall purchase and acquire from Sellers all of Sellers' right, title and interest, free and clear of all Liens (except for the Permitted Liens, if any), in each and all of the Acquired Assets.
- 2.2 <u>Excluded Assets</u>. Notwithstanding anything to the contrary, Buyer shall only acquire the Acquired Assets and neither Buyer nor any Affiliate of Buyer shall acquire, and there shall be excluded from the definition of Acquired Assets, any and all Excluded Assets.
- 2.3 <u>Liabilities to be Assumed by Buyer</u>. Subject to the terms and conditions of this Agreement, at the Closing Buyer shall not assume any Liabilities whatsoever from Sellers except for Assumed Liabilities, if any, that Buyer agrees, in its sole discretion, to assume from Sellers pursuant to the terms of this Agreement and the Sale Order. At the Closing, Sellers shall assign or, pursuant to the Sale Order, be deemed to have assigned, to Buyer, and Buyer shall assume from Sellers and pay when due, perform and discharge, in due course, the Assumed Liabilities, if any. For the avoidance of doubt, all Liabilities that are not Assumed Liabilities shall be Excluded Liabilities.
- 2.4 <u>Excluded Liabilities</u>. Buyer shall not and does not assume, and shall not be obligated to pay, perform, discharge or in any other manner be liable or responsible for any Liabilities other than the Assumed Liabilities, if any.
- 2.5 Receipt of Misdirected Assets; Liabilities. If after the Closing (i) Buyer or any of its Affiliates holds any Excluded Assets or Excluded Liabilities, or (ii) any Seller holds any Acquired Assets or Assumed Liabilities, Buyer or Sellers, as applicable, will promptly transfer (or cause to be transferred) such assets or assume (or cause to be assumed) such Liabilities to or from (as the case may be) the other Party. After the Closing, but prior to any such transfer, the Party receiving or possessing any such asset will hold it in trust for such other Party.

2.6 <u>Assumption and Assignment of Contracts</u>

- (a) Sellers shall assume and, to the extent assignable, assign the Material Contracts listed on Section 2.6(a) of the Disclosure Schedules to Buyer, effective on and as of the Closing (such Contracts, the "Assumed Contracts"); provided, however, Buyer may at any time prior to the Closing, by written notice to Sellers, add or remove Material Contracts from Section 2.6(a) of the Disclosure Schedules, in each case without any adjustment to the Purchase Price.
- Assignment and Assumption Agreement and the Intellectual Property Assignment Agreements, as applicable, sell, and assign to Buyer (the consideration for which is included in the Purchase Price), all Assumed Contracts that may be assigned by any such Seller to Buyer pursuant to sections 363 and 365 of the Bankruptcy Code, as applicable, subject to provision by Buyer of adequate assurance as may be required under section 365 of the Bankruptcy Code and payment by Buyer of the Cure Costs in accordance with Section 6.8 in respect of Assumed Contracts pursuant to and in accordance with section 365 of the Bankruptcy Code, as applicable, and the Sale Order. At the Closing, Buyer shall assume, and thereafter in due course and in accordance with its respective terms (as may be amended) pay, fully satisfy, discharge and perform all of the obligations under each Assumed Contract that are Assumed Liabilities, pursuant to section 365 of the Bankruptcy Code, as applicable.

ARTICLE III. CLOSING; PURCHASE PRICE

3.1 Closing; Transfer of Possession; Certain Deliveries

- (a) The consummation of the Transactions (the "Closing") shall take place on the date of the satisfaction or waiver of all of the conditions set forth in Article 7, but in no event later than the Outside Date unless the Parties mutually agree to a later date pursuant to the terms of this Agreement. The Closing shall be held by electronic exchange of executed documents (or, if the Parties elect to hold a physical Closing, at the offices of DLA Piper LLP (US) set forth in Section 9.4, at 10:00 a.m. Eastern Time, unless the Parties hereto otherwise agree). The actual date of the Closing, effective 12:00 a.m. prevailing Eastern time on such date, is herein called the "Closing Date".
 - (b) At the Closing, Sellers shall deliver to Buyer:
- (i) for each Seller, an officer's certificate, dated as of the Closing Date, executed by a duly authorized officer of such Seller certifying that the conditions set forth in Section 7.1(a) and Section 7.1(b) have been satisfied;
- (ii) counterparts to the Bill of Sale and Assignment and Assumption Agreement, duly executed by each Seller;

- (iii) counterparts to the assignments of Seller Intellectual Property, each of which will be in a form and substance reasonably acceptable to Buyer (the "<u>Intellectual Property Assignment Agreements</u>"), duly executed by each Seller;
- (iv) electronic copies of all books and records in Sellers' possession that relate to the development and manufacturing of the Products, or the Sellers' Amazon Seller Central account, including but not limited to all Regulatory Documentation and all clinical and primary source data;
- (v) for each Seller (or if any Seller is a disregarded entity for U.S. federal income tax purposes, its regarded owner) that is a U.S. Person, an IRS Form W-9 from each such Seller (or, if such Seller is a disregarded entity for U.S. federal income tax purposes, its regarded owner);
- (vi) evidence of consents, authorizations and approvals of the Governmental Authorities and other Persons required as set forth in <u>Section 4.3</u> of the Disclosure Schedules;
- (vii) evidence of all requisite consents or stipulation to the assignment of the Assumed Contracts (including, for the avoidance of doubt, any assignments of Seller IP Licenses, as applicable); and
- (viii) such other instruments of assignment or conveyance duly executed by Sellers as shall be reasonably requested or reasonably necessary to transfer the Acquired Assets to Buyer in accordance with this Agreement.
 - (c) At the Closing, Buyer shall deliver to Sellers:
- (i) instruction to the Escrow Agent to release the Cash Consideration from the Deposit pursuant to the terms of the Escrow Agreement;
- (ii) an officer's certificate, dated as of the Closing Date, executed by a duly authorized officer of Buyer certifying that the conditions set forth in Section 7.2(a) and Section 7.2(b) have been satisfied;
- (iii) a counterpart to the Bill of Sale and Assignment and Assumption Agreement, duly executed by Buyer; and
- (iv) a counterpart to each Intellectual Property Assignment Agreement, duly executed by Buyer.

3.2 Consideration.

(a) <u>Purchase Price; Transaction Value</u>. The aggregate consideration (the "<u>Transaction Value</u>") for the Acquired Assets shall be equal to: (a) (i) an initial credit bid of a portion of Buyer's secured Claims (including claims of its Affiliates) in an amount equal to Three Million Dollars (\$3,000,000.00) (the "<u>Credit Bid Consideration</u>"), <u>plus</u> (ii) an amount in cash equal

to One Hundred Thousand Dollars (\$100,000.00) (the "<u>Cash Consideration</u>") (collectively, the "<u>Purchase Price</u>"), *plus* (b) Buyer's assumption of the Assumed Liabilities.

- (b) <u>Payment of Cash Consideration</u>. On the Closing Date, the Cash Consideration will be released to Sellers from the Deposit at Closing to an account designated by Sellers in writing prior to the Closing.
- satisfy the Credit Bid Consideration by releasing a portion of its then-outstanding secured Claims against Sellers as of the Closing in an amount equal to the Credit Bid Consideration, which amount shall be payable by means of a dollar-for-dollar offset against the amount of such secured Claims pursuant to section 363(k) of the Bankruptcy Code. For the avoidance of doubt, Buyer shall be entitled to a deficiency claim against Sellers that shall be equal to (i) the difference of the amount of the final Credit Bid Consideration and Buyer's (including through its Affiliates) total allowed secured Claims against the Sellers, *minus* (ii) (A) any amounts paid as agreed by Buyer and Sellers for the benefit of Retailers, or (B) any portion of the Retailer Escrowed Funds that have been released to the applicable Retailer beneficiary in accordance with Section 6.1(b).

3.3 Deposit.

- Agent by wire transfer of immediately available funds an aggregate amount in cash equal to One Hundred Thousand Dollars (\$100,000) ("<u>First Deposit Payment</u>"). Upon entry of the Bidding Procedures Order, Buyer shall deposit with the Escrow Agent by wire transfer of immediately available funds an additional aggregate amount in cash equal to One Hundred Thousand Dollars (\$100,000) ("<u>Second Deposit Payment</u>", and together with the First Deposit Payment, the "<u>Deposit</u>"). The Deposit (together with all accrued investment income thereon, if any) shall not be subject to any lien, attachment, trustee process, or any other judicial process of any creditor of any of Sellers.
- (b) The Deposit shall be released and delivered (together with all accrued investment income thereon, if any) by the Escrow Agent to either Buyer or Sellers, as applicable, as follows:
- (i) if the Closing shall occur, (A) One Hundred Thousand Dollars (\$100,000) of the Deposit shall be released to Sellers and applied against the Purchase Price (which, for the avoidance of doubt, will not include any accrued investment income), and (B) One Hundred Thousand Dollars (\$100,000) of the Deposit, plus all accrued investment income on the total sum of the Deposit, if any, shall be released to Buyer;
- (ii) if this Agreement is terminated by Sellers pursuant to <u>Section 8.1(h)</u>, the Deposit, together with all accrued investment income thereon, if any, shall be released to Sellers within five (5) Business Days after such termination; or

- (iii) if this Agreement is terminated for any reason (other than a termination pursuant to <u>Section 8.1(h)</u>), the Deposit, together with all accrued investment income thereon, if any, shall be released to Buyer within five (5) Business Days after such termination.
- Allocation of Purchase Price. The Parties agree to cooperate in the preparation of a schedule (a) allocating the sum of the Purchase Price and the amount of the Assumed Liabilities (to the extent properly taken into account under the Code) among Sellers and (b) the amount allocated to the Acquired Assets sold by each such Seller among such Acquired Assets in accordance with section 1060 of the Code and the Treasury Regulations promulgated thereunder (the "Allocation"). The Parties agree to cooperate in the preparation and filing of an IRS Form 8594 with their Tax Returns consistent with the Allocation. The Parties shall treat and report the Transactions in all respects consistently for purposes of any Tax, including the calculation of gain, loss and basis with reference to the Allocation. The Parties shall not take any action or position inconsistent with the obligations set forth in this Agreement.

ARTICLE IV. REPRESENTATIONS AND WARRANTIES OF SELLERS

Except as set forth in the Disclosure Schedules delivered by the Sellers to Buyer on the Agreement Date, Sellers hereby represent and warrant to Buyer as of the Agreement Date as follows:

- 4.1 <u>Organization</u>. Each Seller (i) is duly qualified or licensed to do business as a foreign corporation and is in good standing under the Laws of each jurisdiction where the nature of the property owned or leased by it or the nature of the Business makes such qualification or license necessary, except where any such failure to be so qualified or licensed, individually in the aggregate, would not result in a Seller Material Adverse Effect; and (ii) pursuant to Sections 1107 and 1108 of the Bankruptcy Code and the Orders of the Bankruptcy Court, has all necessary corporate power and authority to lease the property it operates under lease and to conduct the Business as debtor in possession.
- 4.2 <u>Authorization of Transactions</u>. Each Seller has all requisite corporate power and authority to execute and deliver and, subject to the entry and effectiveness of the Sale Order, to perform its obligations under this Agreement and each of the Transaction Documents to which it is a party. The execution, delivery and performance of this Agreement and such Transaction Documents by Sellers and the consummation of the Transactions by Sellers have been duly and validly authorized and approved by all requisite corporate action of Sellers, and subject to entry of the Sale Order no other corporate proceedings (pursuant to Sellers' organizational documents or otherwise) on the part of any Seller is necessary to authorize the consummation of, and to consummate, the Transactions. Subject to the entry and effectiveness of the Sale Order, this Agreement and each such Transaction Document have been, or at or prior to Closing (as the case may be) will be, duly and validly executed and delivered by Sellers to the extent a party thereto, and, assuming the due authorization, execution and delivery of this Agreement and each such Transaction Document by Buyer, as applicable, constitute (except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or similar Laws now or hereafter in effect relating to creditors' rights generally or general principles of equity or as contemplated

by the Chapter 11 Cases) a valid and binding agreement of Sellers, enforceable against Sellers in accordance with its terms.

- 4.3 <u>Consents and Approvals</u>. Except as set forth in <u>Section 4.3</u> of the Disclosure Schedules and subject to the Sale Order, no consent, approval or authorization of, or declaration, filing or registration with, any Governmental Entity is necessary or required to be made or obtained by Sellers or their Affiliates in connection with the execution, delivery and performance of this Agreement and the Transaction Documents to which a Seller is a party and the consummation of the Transactions, except for the filing of the Intellectual Property Assignment Agreements as applicable and the certain filings required to effectuate the transfer of the Transferred Permits or compliance with any applicable requirements of the Chapter 11 Cases.
- No Conflicts. Subject to the Sale Order having been entered by the Bankruptcy 4.4 Court, the execution, delivery and performance by each Seller of any Transaction Document to which such Seller is (or will become at Closing) a party, and the consummation of the Transactions, does not and will not (a) conflict with or result in any breach of any provision of its certificate of incorporation or bylaws or comparable governing documents, (b) conflict with or result in the breach of the terms, conditions or provisions of or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or give rise to any right of termination, acceleration or cancellation of any right or obligation or to a loss of any benefit under any provision of any Material Contracts, (c) result in a violation of any Law or Order applicable to it or (d) result in the creation or imposition of any Lien on any Acquired Assets other than Permitted Liens, except, in the case of clauses (b), (c) and/or (d), as would not, individually or in the aggregate, be, or reasonably be expected to have a Seller Material Adverse Effect. With regard to the disclosures set forth in Section 4.4 of the Disclosure Schedules, Sellers do not believe such consents are necessary to consummate the Transactions after giving effect to applicable bankruptcy Law. For the avoidance of doubt and notwithstanding anything to the contrary contained herein, such disclosures shall not limit Buyer's termination rights under Section 8.1(g) hereof in any respect.
- 4.5 <u>Acquired Assets</u>. Upon the terms and subject to the conditions contained in this Agreement and subject to requisite Bankruptcy Court approvals and the terms of the Sale Order, at the Closing, subject to any Permitted Liens, Sellers will have good and valid title to, or the right to use, the Acquired Assets. Pursuant, and subject, to the Sale Order, Sellers shall convey such title to or rights to use, all the tangible Acquired Assets, free and clear of all Liens (other than Permitted Liens, if any).
- 4.6 <u>Litigation; Orders</u>. Except as set forth in <u>Section 4.6</u> of the Disclosure Schedules, the Chapter 11 Cases and any adversary proceedings or contested motions commenced in connection therewith, there is no Claim, Proceeding or Order pending, outstanding or, to any Sellers' Knowledge, threatened against any Seller that seeks to restrain or prohibit or otherwise challenge the consummation, legality or validity of the Transactions contemplated hereby.
- 4.7 <u>Books and Records</u>. The books and records maintained with respect to the Business accurately and fairly reflect, in all material respects, the assets and Liabilities of Sellers with respect to the Business.

4.8 <u>Compliance with Laws; Permits</u>. Except as set forth in <u>Section 4.8</u> of the Disclosure Schedules, Sellers are in compliance with all Laws applicable to the Business. Sellers have not received any written notice of or been charged with the violation of any Laws. Sellers have all Permits which are required for the operation of the Business as presently conducted. Sellers are not in default or violation (and no event has occurred which, with notice or the lapse of time or both, would constitute a default or violation) of any term, condition or provision of any Permit to which they are parties.

4.9 Contracts.

- (a) <u>Section 4.9(a)</u> of the Disclosure Schedules sets forth a true and complete list of all executory Contracts of each Seller as of the Agreement Date that, to the Knowledge of Sellers, are material to the Business and/or which relate to any of the Acquired Assets, and Sellers have delivered to Buyer true and complete copies of all such Contracts in the possession of Sellers (the "Material Contracts"), including but not limited to:
- (i) each Contract relating to the Acquired Assets that (i) grants any option, right of first refusal, right of first offer or similar rights to any customer, vendor, supplier, distributor, contractor, collaborator, or other Person or (ii) contains a covenant expressly limiting in any material respect the freedom of any Seller (or that would limit in any material respect the freedom of Buyer after the Closing) to engage in any business with any Person or in any geographic area or to compete with any Person;
- (ii) each contract listed in <u>Section 2.6(a)</u> (Assumed Contracts) of the Disclosure Schedules:
- (iii) any Contract relating to the sale or disposition of Acquired Assets including sale or disposition of Inventory in the Ordinary Course of Business;
- (iv) any Contract providing for the development of any Seller Intellectual Property; and
- (v) any Contract that grants any ownership interest, any license, sublicense or other option or right to any Seller Intellectual Property or by which any Seller is required to grant to any Person any right or license, any covenant not to assert/sue, release or other immunity from suit under or any other rights, to any Seller Intellectual Property.
- (b) True and complete copies of each Material Contract in effect as of the date hereof, including all amendments, modifications and waivers relating thereto, have been made available to Buyer prior to the date hereof. To the Knowledge of Seller, no other party to any Material Contract is in breach of or default under the terms of any such Contract and no event exists which upon notice or the passage of time, or both, would reasonably be expected to (other than as a result of the Chapter 11 Cases and any related bankruptcy filings) (i) give rise to any material default, in the performance by Sellers, or, to the Knowledge of Sellers, by any other party under any of the Material Contracts or (ii) cause or permit the acceleration or other changes of any right or obligation or the loss of any benefit thereunder. Each of the Material Contracts is a legal,

valid, binding and enforceable obligation of Sellers and, to the Knowledge of Sellers, of each other party thereto, and is in full force and effect. Sellers have not received any written notice of any termination, default or event that with notice or lapse of time, or both, would constitute a default by any Seller under any Material Contract.

4.10 Real Property. No Seller owns any real property.

4.11 Intellectual Property.

- (a) Section 4.11(a) of the Disclosure Schedules sets forth a true and complete list, as of the date hereof, of: (i) all Patents, (ii) all pending and registered Trademarks; (iii) all pending and registered Copyrights; (iv) all domain names and social media accounts (items (i) to (iii) collectively referred to as the "Seller IP Registrations"); and (v) all material unregistered Trademarks and software, in each case included in the Seller Owned Intellectual Property and including the jurisdiction in which each item has been registered or filed, application or serial number or similar identifier, the filing date, the applicable issuance, registration or grant date, as applicable. The Seller Owned Intellectual Property is valid, enforceable, and, with respect to the Seller IP Registrations, subsisting.
- (b) All necessary filing, examination, registration, maintenance, annuity and renewal fees due or actions required (including proofs and working or use) in connection with the Seller Intellectual Property listed or required to be listed in Section 4.11(a) of the Disclosure Schedules and having a due date on or before the date hereof, have been paid or taken, as the case may be, and except as set forth in Section 4.11(b) of the Disclosure Schedules, none of the Seller Intellectual Property is subject to any filing, examination, registration, maintenance, annuity and renewal fees falling due or actions required (including proofs of working or use) at the time of Closing.
- (c) Sellers exclusively own and possess, free and clear of all Liens, all right, title and interest in and to, or have the right to use pursuant to a valid and enforceable written license that is set forth in Section 4.9(a)(v) of the Disclosure Schedules (the "Seller IP Licenses"), all Intellectual Property used in, material to or otherwise necessary for the operation of the Business as currently conducted and as proposed to be conducted as of the Closing Date. Sellers have taken all reasonable or necessary actions and follows all practices common in the industry to maintain, protect and enforce the Seller Owned Intellectual Property, including the secrecy, confidentiality and value of its trade secrets and other confidential information. The Seller Owned Intellectual Property, together with the Intellectual Property licensed under the Seller IP Licenses, are sufficient for the operation of the Business as currently conducted. Immediately following the Closing, Buyer will have identical rights in all Seller Intellectual Property and other Intellectual Property used in the Business as those possessed by the Sellers prior to the Closing.
- (d) Seller has complied with all applicable Law in filing and prosecuting each item of the Seller Intellectual Property owned or licensed by Sellers and all filings, payments and other actions required to be made or taken to maintain each item thereof in full force and effect have been made by the applicable deadline.

- (e) The operation of the Business does not infringe, misappropriate or otherwise violate, and has not infringed, misappropriated or otherwise violated, any Intellectual Property of any other Person. No interference, opposition, reissue, reexamination or other Proceeding of any nature is or has been pending or, to the Knowledge of Sellers, threatened, in which the scope, validity or enforceability of any Seller Intellectual Property is being, has been or could reasonably be expected to be contested or challenged, and no Proceedings are pending or, to the Knowledge of Sellers, threatened which allege that a Seller is infringing, misappropriating, diluting or otherwise violating the Intellectual Property of any Person. To the Knowledge of Sellers, no Person has infringed, misappropriated or otherwise violated, or is infringing, misappropriating or otherwise violating, any Seller Owned Intellectual Property. Sellers have not received any written charge, complaint, claim, demand, notice or other written communication or notice from any Person claiming that the operation of the Business infringes, misappropriates, or violates any Intellectual Property of any Person.
- (f) Except as set forth in <u>Section 4.11(f)</u> of the Disclosure Schedules, each of the past and present employees of, and consultants to, the Sellers who has developed, created, invented or authored any Intellectual Property used in, material to or otherwise necessary for the operation of the Business has entered into agreements with the applicable Seller pursuant to which such employee or consultant (i) agrees to protect the such Seller's confidential information and trade secrets, and (ii) assigns to such Seller all Intellectual Property created or otherwise developed by such employee or consultant in the course of his, her or its relationship with such Seller, without further consideration or any restrictions or obligations on the use or ownership of such Intellectual Property.
- 4.12 <u>Brokers' Fees and Commissions</u>. Except as set forth on <u>Section 4.12</u> of the Disclosure Schedules, no agent, broker, Person or firm acting on behalf of any Seller or under any Seller's authority is or will be entitled to any advisory, commission or broker's or finder's fee or commission from any of the parties hereto in connection with any of the Transactions.
- 4.13 <u>Financial Statements</u>. The consolidated balance sheets and the related consolidated statements of comprehensive loss, stockholders' equity and cash flows (including, in each case, any related notes and schedules thereto) of the Company filed since January 1, 2021 with the SEC have been prepared in conformity with GAAP (except, in the case of unaudited statements, as permitted by Form 10-Q of the SEC) applied on a consistent basis during the periods involved (except as otherwise noted therein or to the extent required by GAAP) and present fairly, in all material respects, the financial position and the results of operations and cash flows of the Sellers as of the dates or for the periods presented therein (subject, in the case of unaudited statements, to normal year-end adjustments).

4.14 Regulatory Matters.

(a) Sellers have made available to Buyer true, correct and complete copies of all: (A) information about adverse drug experiences obtained or otherwise received by Sellers from any source pertaining to the Products; and (B) Regulatory Documentation and Regulatory Authorizations from or with the FDA and all other applicable Regulatory Authorities filed, submitted, exchanged, or held by Sellers relating to the Product or any other Seller Intellectual

Property or necessary or useful to conduct the Business. All such Regulatory Authorizations, if any, are (i) in full force and effect, (ii) validly registered and on file with applicable Regulatory Authorities, (iii) in compliance with all material filing and maintenance requirements, and (iv) in good standing, valid and enforceable. Sellers have at all times fulfilled and performed all of their material obligations with respect to such Regulatory Authorizations, and no event has occurred, and no fact, circumstance or condition exists, which allows, or after notice or lapse of time would allow, lapse, revocation, or termination thereof. Subject to Sellers and Buyer each submitting transfer letters to the FDA or other applicable Governmental Authority, the rights and benefits of each Seller Regulatory Submission will be available to Buyer on the Closing Date on terms substantially identical to those enjoyed by Sellers immediately prior to the Closing Date.

- (b) (i) Sellers have submitted to and timely filed with the FDA all required notices, declarations, supplemental applications and annual or other reports or documents, including adverse event reports and field alerts, with respect to Product Regulatory Submissions and commercialization activities, (ii) Sellers have paid all fees required by any Governmental Authority with respect to Product Regulatory Submissions, including filing fees applicable to the Products and with respect to all Permits, and (iii) Sellers are in compliance in all material respects with all applicable Health Laws that affect the Acquired Assets, the Assumed Liabilities or the Business and all terms and conditions of the Regulatory Authorizations.
- (c) All clinical trials conducted or being conducted with respect to the Products by or at the direction of Sellers have been and are being conducted in compliance in all respects with the required experimental protocols, procedures and controls and in all material respects with all applicable Health Laws. No clinical trial conducted by or, on behalf of, Sellers have been terminated or suspended by any Regulatory Authority and Sellers have not received any notification or other communication from any institutional review board, ethics committee or safety monitoring committee raising any issues that may result in a clinical hold or otherwise delay, materially restrict or otherwise limit or impair the use of any clinical studies proposed or currently conducted by, or on behalf of, Sellers, or in which Sellers have participated and, to the Knowledge of Sellers, no such action has been threatened.
- (d) None of Sellers or any Person acting on Sellers' behalf has, with respect to the Product, (i) been subject to a Regulatory Authority shutdown or import or export prohibition or (ii) received any oral notice, FDA Form 483, or other written Regulatory Authority notice of inspectional observations, "warning letters," "untitled letters" or written demand or written request to make any change to any Product or any processes or procedures, or any similar correspondence from any Regulatory Authority alleging or asserting non-compliance with any applicable Health Law or Regulatory Authorization and, to the Knowledge of Seller, no Regulatory Authority is considering such action. Sellers have not received any written or oral notice or other communication from any Regulatory Authority, as applicable, withdrawing, terminating, suspending or placing any Regulatory Submission or Regulatory Authorization applicable to the Product on hold. There are no investigations, suits, Claims, actions or proceedings pending or threatened which allege any violation by any Sellers or any third party engaged by Sellers with respect to the Product of any Health Law. No Product has been seized, withdrawn, recalled, detained or subject to a suspension of manufacturing and there are no facts or circumstances

reasonably likely to cause the seizure, denial, withdrawal, recall, detention, field notification, field correction, safety alert or suspension of manufacturing relating to any such product. No proceedings in the United States or any other jurisdiction seeking the withdrawal, recall, correction, suspension, import detention or seizure of any such Company product are pending or threatened against the Company.

- (e) Sellers have not used in any capacity the services of any Person debarred, excluded, or disqualified under 21 U.S.C. Section 335a, 42 U.S.C. Section 1320a-7, 21 C.F.R. Section 312.70, or other similar Laws. Neither Sellers nor any individual employed by Sellers has been disqualified or debarred by any Governmental Entity, including but not limited to the FDA pursuant to the FD&C Act, or been convicted of any crime under any Health Law or has engaged in any conduct that has resulted, or would reasonably be expected to result, in debarment, exclusion, or disqualification under 21 U.S.C. Section 335a, 42 U.S.C. Section 1320a-7, 21 C.F.R. Section 312.70, or any similar Laws or engaged in any misconduct relating to the development, approval, marketing or sale of pharmaceutical products, including the Products.
- (f) Neither Sellers, nor any Person acting on behalf of Sellers, has made an untrue statement of a material fact or fraudulent statement to the FDA or other Governmental Authority, or failed to make a statement or committed an act in contravention to the policies of FDA or any other Governmental Entity.
- (g) No Product is reimbursed (directly or indirectly, in whole or in part) by a Third Party Payor.

4.15 <u>Taxes</u>.

- (a) (i) All income and other material Tax Returns required to be filed with respect to the Business, the Acquired Assets and Assumed Liabilities have been timely filed with the appropriate taxing authority in all jurisdictions in which such Tax Returns are required to be filed (after giving effect to any valid extensions of time in which to make such filings), and all such Tax Returns are true, complete and correct in all material respects; and (ii) all material Taxes payable with respect to the Business, the Acquired Assets and Assumed Liabilities, whether or not shown on any Tax Return, have been timely paid.
- (b) No Claim, audit or other proceeding with respect to any Taxes or Tax Returns with respect to the Business, the Acquired Assets or Assumed Liabilities is currently in progress or has been proposed or threatened in writing.
- (c) Sellers have not received written notice of any Tax deficiency outstanding, proposed or assessed, nor have Sellers executed any waiver of any statute of limitations in respect of Taxes nor agreed to any extension of time with respect to a Tax assessment, collection or deficiency, in each case, with respect to the Business, the Acquired Assets or Assumed Liabilities.
 - (d) There are no material liens for Taxes upon any of the Acquired Assets.

- (e) No Seller has any obligation for Taxes pursuant to any Assumed Contract (other than an Assumed Contract entered into in the Ordinary Course of Business, the primary subject matter of which is not Taxes) that Buyer is assuming as a result of the Transactions.
- (f) Each Seller has withheld and paid all material Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder, or other third party, and complied in all material respects with any information reporting or backup withholding requirement with respect thereto. Each Person providing services to any Seller has been properly classified as an employee or independent contractor, as the case may be, for all Tax purposes and with respect to all employee benefit plans.
- (g) Each Seller has collected all material sales, use and value added taxes required to be collected, and has remitted, or will remit on a timely basis, such amounts to the appropriate Taxing Authority and has collected and retained properly completed exemption certificates for all exempt transactions.
- (h) None of the Acquired Assets constitutes stock, partnership interests or any other equity interest in any Person for U.S. federal income Tax purposes.
- 4.16 <u>Insurance</u>. <u>Section 4.16</u> of the Disclosure Schedules sets forth with respect to the Business a complete and accurate list of the insurance policies of Sellers as of the date hereof. All insurance policies set forth on <u>Section 4.16</u> of the Disclosure Schedules are in full force and effect and are valid and enforceable and Sellers are not in material breach of or default under any such insurance policies. Sellers have not taken any action or failed to take any action that (with or without notice or lapse of time, or both), would constitute such a breach or default or permit termination or modification of any of the insurance policies. All premiums due thereunder have been paid. There are no material claims under any of the insurance policies set forth on <u>Section 4.16</u> of the Disclosure Schedules for which coverage has been denied or disputed by the applicable insurance carrier (other than a customary reservation of rights notice).
- 4.17 <u>Interested Party Transactions</u>. Except as set forth in <u>Section 4.17</u> of the Disclosure Schedules, there are no Contracts between any Seller, on the one hand, and any (i) any employee, officer, director, direct or indirect equity holder, or Affiliate of such Seller; (ii) any employee, officer, director, direct or indirect equity holder, or Affiliate of any of the foregoing; or (iii) any family member of any of the foregoing natural Persons, on the other hand, other than in their capacity as employees or contractors of such Seller in the Ordinary Course of Business.
- 4.18 <u>Disclosure</u>. Sellers have made available to the Buyer all the information reasonably requested from the Sellers that Buyer has requested for deciding whether to acquire the Acquired Assets. To Sellers' Knowledge, no representation or warranty of the Sellers contained in this Agreement, as qualified by the Disclosure Schedules, and no certificate furnished or to be furnished to Buyer at the Closing contains any untrue statement of a material fact or, to the Sellers' Knowledge, omits to state a material fact necessary in order to make the statements contained herein or therein not misleading in light of the circumstances under which they were made.

REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer hereby represents and warrants to Sellers, as of the Agreement Date, and as of the Closing, as follows:

- 5.1 <u>Organization</u>. Buyer is duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization. Buyer has all necessary limited liability company power and authority to own and operate its assets and to conduct its business.
- 5.2 <u>Due Authorization, Execution and Delivery; Enforceability</u>. Buyer has all requisite limited liability company power and authority to execute and deliver this Agreement and the other Transaction Documents to which it is (or will become at Closing) a party and to perform its obligations hereunder and thereunder. The execution, delivery and performance by Buyer of this Agreement and the other Transaction Documents to which it is (or will become at Closing) a party and the consummation of the Transactions have been duly and validly authorized by all requisite limited liability company action on the part of Buyer and no other limited liability company action on the part of Buyer is necessary to authorize this Agreement and such other Transaction Documents and to consummate the Transactions (subject, in the case of the obligation to consummate the Transactions, to the entry of the Sale Order). This Agreement and the other Transaction Documents to which Buyer is (or will become at Closing) party have been (or will be) duly and validly executed and delivered by Buyer and (assuming the due authorization, execution and delivery by all Parties hereto and thereto, other than Buyer) constitute (or will constitute) valid and binding obligations of Buyer enforceable against Buyer in accordance with their terms (subject to the entry of the Sale Order), in each case except as enforceability may be limited by applicable bankruptcy, insolvency or similar Laws affecting the enforcement of creditors' rights generally or by equitable principles relating to enforceability.
- 5.3 Governmental Approvals. No notice to, consent, approval or authorization of or designation, declaration or filing with any Governmental Entity is required by Buyer with respect to Buyer's execution and delivery of any Transaction Document to which it is (or will become at Closing) a party or the consummation of the Transactions, except (a) the Sale Order having been entered by the Bankruptcy Court and (b) any consent, approval or authorization of or designation, declaration or filing with any Governmental Entity the failure of which to be made or obtained would not, individually or in the aggregate, be reasonably expected to materially impair the Buyer's ability to consummate the Transaction.
- 5.4 No Conflicts. The execution, delivery and performance by Buyer of any Transaction Document to which Buyer is (or will become at Closing) a party and the consummation of the Transactions, does not and will not (a) conflict with or result in any breach of any provision of its certificate of formation or operating agreement or comparable governing documents, (b) conflict with or result in the breach of the terms, conditions or provisions of or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or give rise to any right of termination, acceleration or cancellation under, any material Contract of Buyer, or (c) result in a violation of any Law or Order applicable to it, except, in the case of clauses (b) and (c), as would not, individually or in the aggregate, materially impair the Buyer's ability to consummate the Transaction.

- 5.5 <u>Sufficiency of Funds</u>. Buyer has sufficient cash on hand to enable it to consummate the Transactions contemplated by this Agreement, including the timely satisfaction of the Assumed Liabilities (including any Cure Costs).
- 5.6 <u>Adequate Assurances Regarding Executory Contracts</u>. Buyer is and will be capable of satisfying the conditions contained in section 365 of the Bankruptcy Code with respect to the Assumed Contracts.
- 5.7 <u>Exclusive Representations and Warranties</u>. Except for the representations and warranties contained in this <u>Article 5</u>, none of Buyer, its Affiliates, nor any of their respective Representatives, makes or has made any other representation or warranty on behalf of Buyer. Except for the representations and warranties contained in this <u>Article 5</u>, Buyer disclaims all liability and responsibility for any representation, warranty, projection, forecast, statement, or information made, communicated, or furnished (orally or in writing) to Sellers or their Representatives (including any opinion, information, projection, or advice that may have been or may be provided to Sellers by any Representative of Buyer or any of their respective Affiliates).
- 5.8 As Is and Where Is. BUYER HEREBY ACKNOWLEDGES AND AGREES THAT, EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN <u>ARTICLE 4</u> (AS MODIFIED BY THE DISCLOSURE SCHEDULES), SELLERS MAKE NO REPRESENTATIONS OR WARRANTIES WITH RESPECT TO ANY MATTER RELATING TO THE ACQUIRED ASSETS. WITHOUT IN ANY WAY LIMITING THE FOREGOING, SELLERS HEREBY DISCLAIM ANY WARRANTY, EXPRESS OR IMPLIED, OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE AS TO ANY PORTION OF THE ACQUIRED ASSETS. BUYER WILL ACCEPT THE ACQUIRED ASSETS AT THE CLOSING "AS ISWHERE IS".

ARTICLE VI. COVENANTS OF THE PARTIES

- 6.1 <u>Interim Operating Covenants</u>. During the period from the Agreement Date and continuing until the earlier of the termination of this Agreement in accordance with its terms or the Closing (the "<u>Interim Period</u>"), except as may be required by Order of the Bankruptcy Court (provided that no Seller directly or indirectly petitioned, sought, requested or moved for such order of the Bankruptcy Court or authorized, supported or directed any other Person to petition, seek, request or move for such Order of the Bankruptcy Court), Sellers shall carry on the Business in the Ordinary Course of Business and, to the extent consistent therewith, use commercially reasonable efforts to preserve the Business intact and preserve the goodwill of and relationships with Governmental Entities, customers, suppliers, employees and others having business dealings with the Business.
- (a) Notwithstanding the first sentence of <u>Section 6.1</u>, during the Interim Period, Sellers shall not, and shall cause their Affiliates not to, directly or indirectly, without the prior written consent of Buyer:

- (i) modify, amend, terminate, waive any rights or obligations under or otherwise seek to reject any Assumed Contract or any material contract to which an Acquired Company is a party;
- (ii) lease, license, surrender, relinquish, sell, transfer, convey, assign or otherwise dispose of any interest in any Acquired other than (i) Inventory and immaterial assets, in each case in the Ordinary Course of Business, (ii) pursuant to an Order of the Bankruptcy Court (provided that no Seller directly or indirectly petitioned, sought, requested or moved for such order of the Bankruptcy Court or authorized, supported or directed any other Person to petition, seek, request or move for such Order of the Bankruptcy Court);
- (iii) mortgage, pledge or subject to Liens (other than Permitted Liens) any of the Acquired Assets;
- (iv) except as required by applicable Law, (i) increase any element of compensation or any benefit of any current or former director or independent contractor of any Seller or any Subsidiary of any Seller, or any officer or employee; (ii) accelerate the vesting or payment of the compensation payable or the benefits provided or to become payable or provided to any current or former director or independent contractor of any Seller or any Subsidiary of any Seller, or any officer or employee, or otherwise pay any amounts not due to any such individual (including with respect to severance); or (iii) adopt, enter into, amend, terminate or renew any Benefit Plan (including any employment, severance, consulting or other individual agreement), or which would otherwise increase the Assumed Liabilities, or adopt or enter into any other plan, program, policy, agreement or arrangement that would be considered a Benefit Plan if it were in existence on the date of this Agreement that is or becomes an Assumed Benefit Plan, or which would otherwise increase the Assumed Liabilities:
- (v) adopt a plan or agreement of complete or partial liquidation or dissolution or otherwise fail to maintain its limited liability company, corporate or similar existence;
- (vi) institute, settle or agree to settle any material Proceeding (other than the Chapter 11 Cases) before any Governmental Entity relating to the Acquired Assets or the Business, or modify in any manner that is adverse to the Business or the Acquired Assets, rescind, reject or terminate a transferrable Permit (or application therefor) relating to the Acquired Assets;
- (vii) with respect to any Acquired Asset: (A) make, change or revoke any material Tax election, (B) change any annual Tax accounting period, (C) enter into any agreement with respect to material Taxes or apply for any Tax ruling, (D) file any amended Tax Return (unless otherwise required to do so under applicable Law), (E) settle or compromise any Tax Liability or Claim for a Tax refund, (F) surrender any right to Claim a refund for Taxes, (G) incur any material Tax Liability outside of the Ordinary Course of Business, (viii) file any Tax Return other than one prepared in a manner consistent with past practice, (H) consent to an extension of the statute of limitations applicable to any Tax Claim or assessment (other than pursuant to extensions of time to file Tax Returns obtained in the Ordinary Course of Business) or (I) change (or request any Governmental Entity to change) any aspect of its method of accounting for Tax purposes;

- (viii) fail to maintain in full force and effect insurance policies covering the Acquired Assets and the operation of the Business, in form and amount consistent with past practice;
 - (ix) amend any organizational documents of Sellers;
- (x) (A) authorize, issue or grant, (B) pledge, encumber or permit the creation of, or suffer to exist, any Lien (other than a Permitted Lien) on, or (C) sell, transfer, assign, convey, distribute or otherwise dispose of (including by merger with any other Person), any stock, partnership interest or joint venture interest or other equity ownership interest of Sellers or any Affiliate:
- (xi) incur any capital expenditures or any Liabilities in respect thereof other than in the Ordinary Course of Business;
- (xii) settle, compromise, or otherwise satisfy any Liability with an asserted amount in excess of \$500,000;
- (xiii) create, incur, assume, guarantee or otherwise be liable with respect to any debt (or amend or modify terms of any such debt) except as does not relate to any Acquired Company;
- (xiv) make any loans, advances or capital contributions to, or investments in, any Person; or
 - (xv) enter into any Contract to do any of the foregoing;
- (b) Notwithstanding the foregoing, during the Interim Period, Sellers affirmatively covenant and agree as follows (collectively, the "Interim Operating Covenants"):
- (i) To the extent that the following actions have not already been authorized by Sellers pursuant to the IOI, effective as of the Agreement Date through the Interim Period, Sellers hereby authorize Buyer to (i) markdown all VAZALORE® Products (except for VAZALORE® 325 mg Products, unless and until Sellers provide further written consent) with an expiry date of September 30, 2023 or earlier, at all Retailers, (ii) authorize returns for all Products with an expiry date range of February 1, 2023 through May 31, 2023 at all Retailers, (iii) negotiate Product promotions with all Retailers, (iv) manage the pricing levers with respect to Product sales on the Seller's Amazon Seller Central account, and (v) administer the return process in respect of the previously announced discontinuance of VAZALORE® 81mg 12-count Products at applicable Retailers, provided that (A) with regard to the foregoing clauses (b)(i) through (b)(v), Buyer shall consult with Sellers, and (B) with regard to the foregoing clause (b)(v), insofar as such actions could create incremental Liabilities against Sellers in excess of Buyer's aggregate return liability to Retailers amount as of March 31, 2023, Buyer and Sellers shall reasonably agree on such steps prior to undertaking them.

- (ii) On the Agreement Date, contemporaneous with Buyer's funding of the Deposit set forth in Section 3.3(a), Seller transfer \$825,000 in cash to Buyer in respect of the following designated Retailer liabilities: (A) for the benefit of Target—\$500,000, which amount shall constitute full and final satisfaction of that certain invoice number 12123SC1027, dated January 24, 2023, as previously agreed by and between Sellers and Buyer in Q3 2022; and (B) for the benefit of Walgreens—\$325,000, which amount shall be allocated in respect of that certain CLR arrangement previously negotiated as between Seller and Walgreens.
- shall deposit \$900,000 into an escrow account acceptable to Buyer, to be held in escrow solely for the benefit of the following Retailers: Wal-Mart, Walgreens, CVS, Rite-Aid, and certain of Sellers' grocery store Product retailers (collectively, the "Retailer Escrowed Funds") in relation to amounts that are or will become due and payable during the Interim Period for Product returns required under certain discontinuation letter, dated December 1, 2022, issued by Sellers, in relation to VAZALORE® 81mg 12-count Products. Such escrow arrangement shall be acceptable to Buyer in its sole discretion. During the Interim Period, Sellers and Buyer shall use their reasonable best efforts to reconcile all such amounts due and payable to applicable Retailers. No later than 180 days following the Closing Date, the Parties shall complete the reconciliation of such amounts and any surplus amounts from the Retailer Escrowed Funds shall be promptly returned to Sellers or their designee(s).
- (iv) During the Interim Period, Sellers agree Buyer is authorized and entitled to continue to withhold any cash payments to Sellers in accordance with the terms of to that certain Warehousing and Distribution Services Agreement, dated as of July 1, 2021, by and between Sellers and Buyer's affiliate, or otherwise, and Buyer's Claim amount shall be reduced on a dollar-for-dollar basis on account of such net withholdings, if any.
- (v) If for any reason this Agreement is terminated prior to the Closing of the Transactions contemplated by this Agreement, this <u>Section 6.1(b)</u> shall survive such early termination until all obligations of the Parties hereunder have been performed or discharged.
- (c) Sellers acknowledge and agree that their commercial arrangements with Buyer and its Affiliates have been modified by the IOI and this Agreement, including in respect of the Interim Operating Covenants set forth in this Section 6.1. Such covenants are consistent with, and shall continue in, the Ordinary Course of Business of Sellers. Furthermore, Sellers acknowledge and agree that Buyer's and its Affiliates' continued performance in reliance upon and relation to such covenants does not implicate the automatic stay under section 362 of the Bankruptcy Code in any respect.
- 6.2 Access. Subject to applicable Law, during the Interim Period, Sellers shall give Buyer and its Representatives reasonable access during normal business hours to the offices, properties, officers, employees, accountants, auditors, counsel and other representatives, data, books and records of Sellers to the extent relating to the Acquired Assets, as Buyer reasonably deems necessary in connection with effectuating the Transactions contemplated by this Agreement. It is acknowledged and understood that no investigation by Buyer or other information received by Buyer shall operate as a waiver or otherwise affect any representation, warranty or

other agreement given or made by Sellers hereunder. Buyer agrees that any on-site inspections of any of Acquired Assets, shall be conducted in the presence of Sellers or their Representatives. All inspections shall be conducted so as not to interfere unreasonably with the use of any of the Acquired Assets by Sellers, or operation of the Business, and shall not violate any applicable Law or confidentiality obligations of any Seller. Notwithstanding the foregoing, Buyer and its Representatives shall not be entitled to any records or information pursuant to this Section 6.2, or otherwise, that is subject to legal privilege or that would or could trigger a breach or violation of any obligations under any confidentiality or privacy provision or privacy rule, to which any Seller is subject by Law or Contract.

6.3 <u>Public Announcements</u>. Buyer and Sellers will consult with each other before issuing, and provide each other the opportunity to review and comment upon, any press release or public announcement of this Agreement and the Transactions, but neither Buyer nor Sellers shall issue any press release without the prior written approval of the other Party, which approval shall not be unreasonably withheld, in each case except as may be required by Law, court process (including the filing of this Agreement with the Bankruptcy Court as an exhibit to the Bidding Procedures Motion) or by obligations pursuant to any listing agreement with any national securities exchange, in which case the non-disclosing party will have the right to review and comment on such release, announcement or communication prior to publication. Buyer and Sellers shall cause their respective Affiliates and Representatives to comply with this <u>Section 6.3</u>. This section shall survive the Closing or earlier termination of this Agreement.

6.4 Tax Matters.

- (a) Half of the Transfer Taxes arising out of the transfer of the Acquired Assets pursuant to this Agreement shall be borne by Sellers, with the other half being borne by Buyer. Sellers and Buyer shall cooperate to timely prepare and file any Tax Returns relating to such Transfer Taxes, including any Claim for exemption or exclusion from the application or imposition of any Transfer Taxes. Buyer or Sellers, as applicable, shall file all necessary documentation and returns with respect to such Transfer Taxes when due, and shall promptly, following the filing thereof, furnish a copy of such return or other filing and a copy of a receipt showing payment of any such Transfer Tax to the other Parties hereto. Buyer and Sellers shall each pay half of the Transfer Taxes when due.
- (b) Each of Buyer, on the one hand, and Sellers, on the other hand, shall cooperate fully, as and to the extent reasonably requested, in connection with the preparation and filing of Tax Returns and any audit, litigation or other proceeding with respect to Taxes and shall furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information and assistance relating to the Acquired Assets and the Business as is reasonably necessary for filing of all Tax Returns, including any Claim for exemption or exclusion from the application or imposition of any Taxes, the preparation for any audit by any Tax Authority and the prosecution or defense of any Proceeding relating to any Tax Return.

6.5 Commercially Reasonable Efforts.

- (a) Unless otherwise specifically set forth in this Agreement, each Party shall use its commercially reasonably efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other Parties in doing, all things necessary, proper or advisable under applicable Law to consummate and make effective the Transactions. Without limiting the generality of the foregoing, the Parties will use their respective reasonable best efforts to (i) take all actions necessary to transfer the Acquired Assets, (ii) take all actions necessary to cause all conditions set forth in Article 7 to be satisfied as soon as practicable, (iii) lift or rescind any existing Order preventing, prohibiting or delaying the consummation of the Transactions, (iv) effect all necessary registration, applications, notices and other filings required by applicable Law, including, as applicable to Sellers, under the Bankruptcy Code, and (v) execute and deliver any additional instruments necessary to fully carry out the purposes of this Agreement. Buyer shall not and shall cause its Affiliates and Representatives not to take any action that would reasonably be expected to prevent or materially delay the approval of any Governmental Entity of any of the filings referred to in this Section 6.5.
- (b) To the extent Buyer comes into the possession of any Personally Identifiable Information, Buyer shall immediately deliver such information to Sellers and destroy and/or delete all copies of such information and provide Sellers with evidence of such destruction.

6.6 <u>Further Assurances</u>.

- Subject to the other provisions hereof, Sellers shall use their reasonable best (a) efforts to perform its obligations hereunder and to take, or cause to be taken, and do, or cause to be done, all things necessary, proper or advisable under applicable Law to cause the Transactions contemplated herein to be effected as soon as practicable, but in any event on or prior to the Outside Date, in accordance with the terms hereof and shall cooperate in a commercially reasonable manner with each other Party hereto and its representatives in connection with any action required to be taken as a part of its obligations hereunder. From time to time after the Closing, each of Buyer and Sellers, at the request of the other, will execute and deliver such other instruments of conveyance and transfer or other instruments or documents, and take or arrange for such other actions, as may reasonably be required to effect any of the Transactions contemplated by this Agreement; provided that, notwithstanding anything to the contrary in this Section 6.6 or any other provision of this Agreement, neither Buyer nor Sellers shall (i) be required to execute any document or take any action that would increase the Liability or obligation of the Party of whom such document or action is requested beyond that such Party would have pursuant to the other provisions of this Agreement, (ii) require or cause the Party of whom such action or document is requested to initiate, join in or otherwise become a Party to any Legal Proceeding, or (iii) cause such Party to incur any material cost or expense that is not already imposed upon it by another provision of this Agreement; provided, further that Sellers' obligations under this Section 6.6 shall terminate upon the effective date of a confirmed plan of liquidation, or conversion or dismissal of the Bankruptcy Case.
- (b) Sellers will promptly notify Buyer in writing of: (i) any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the Transactions contemplated by this Agreement; (ii) any notice or other communication from any Governmental Entity, or any Legal Proceeding by any Governmental

Entity, related to or in connection with the Transactions contemplated by this Agreement (including that may restrain, enjoin or otherwise prohibit the consummation of the Transactions contemplated by this Agreement); (iii) the discovery of any variances from, or the existence or occurrence of any event, fact or circumstance arising after the execution of this Agreement that would reasonably be expected to cause, any of the representations and warranties contained herein to be untrue or inaccurate; and (iv) any event that has had, or is reasonably expected to have, a Seller Material Adverse Effect, or would otherwise cause, or reasonably be expected to cause, the failure of any condition to Closing for the benefit of Buyer.

- (c) From and after the Closing, if any Seller or any of its Affiliates receives any right, property or asset that is a Acquired Asset, Sellers shall promptly transfer or cause their Affiliate to transfer such right, property, or asset (and shall promptly endorse and deliver any such asset that is received in the form of cash, checks or other documents) to Buyer, and such asset will be deemed the property of Buyer held in trust by Sellers for Buyer until so transferred.
- (d) With respect to any Acquired Asset whose title is not transferred to Buyer as of the Closing, Sellers hereby grant to Buyer an irrevocable, exclusive, royalty-free, fully paid-up, sublicensable, transferrable license for no additional consideration.
- (e) For the avoidance of doubt and notwithstanding anything in this Agreement to the contrary: (i) Sellers acknowledge and agree that, from and after the Closing, Sellers shall not have any right, title or interest in or to any Seller Intellectual Property; and (ii) from and after the Closing, Sellers shall cease and discontinue any and all use of any and all Seller Intellectual Property.

6.7 <u>Bankruptcy Court Matters</u>.

- (a) Without limiting any other covenant set forth herein in any respect, Sellers shall use commercially reasonable efforts to provide draft copies of all motions, including "first day" motions, applications, and other documents that any Seller intends to file with the Bankruptcy Court in the Chapter 11 Cases to counsel for Buyer at least two (2) Business Days prior to the date when such Person intends to file such document, and shall consult in good faith with such counsel regarding the form and substance of any such proposed filing.
- Order. Sellers shall use commercially reasonable efforts to comply (or obtain an order from the Bankruptcy Court waiving compliance) with all requirements under the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules and the Local Bankruptcy Rules in connection with obtaining approval of the Bidding Procedures Order and the Sale Order. Sellers shall consult with Buyer and its Representatives concerning the Bidding Procedures Order, the Sale Order, any other Orders of the Bankruptcy Court relating to the Transactions, and the bankruptcy proceedings in connection therewith, and use commercially reasonable efforts to provide Buyer with copies of applications, pleadings, notices, proposed Orders and other documents relating to such proceedings as soon as reasonably practicable prior to filing. In furtherance of the foregoing, Sellers shall provide Buyer with a reasonable opportunity to review and comment on all material

motions to be filed in the Chapter 11 Cases that relate to the transactions contemplated by this Agreement, to the extent practicable, prior to their filing with the Bankruptcy Court

- (c) Buyer agrees that it will promptly take such actions as are reasonably requested by Sellers to assist in obtaining entry of the Bidding Procedures Order and the Sale Order and a finding of adequate assurance of future performance by Buyer, including furnishing affidavits or other documents or information for filing with the Bankruptcy Court for the purposes, among others, of providing necessary assurances of performance by Buyer under this Agreement and demonstrating that Buyer is a "good faith" purchaser under section 363(m) of the Bankruptcy Code; provided, however, in no event shall Buyer or Sellers be required to agree to any amendment of this Agreement.
- (d) Sellers shall satisfy each of the following requirements by the deadlines set forth herein, each of which may be extended or waived with Buyer's written consent in its sole discretion (collectively, the "Milestones"):
- (i) On <u>the Agreement Date</u>, Sellers shall have made the payments to or for the benefit of certain Retailers in accordance with <u>Section 6.1(b)(ii)</u> hereof;
- (ii) No later than <u>one business day</u> before the Petition Date, Sellers shall have funded an escrow account with the Retailer Escrowed Funds in accordance with <u>Section</u> <u>6.1(b)(iii)</u> hereof;
- (iii) Sellers shall have commenced the Chapter 11 Cases by no later than **April 13, 2023**;
- (iv) No later than <u>the Petition Date</u>, Sellers shall have filed the Bidding Procedures Motion, the form and substance of which shall be acceptable to Buyer;
- (v) No later than <u>21 days after the Petition Date</u> Sellers shall obtain entry of the Bidding Procedures Order by the Bankruptcy Court, the form and substance of which shall be acceptable to Buyer, which approves the Bidding Procedures Motion, the Bidding Procedures, the Bid Protections, and designates Buyer as the "Stalking Horse";
- (vi) No later than <u>40 days after the Petition Date</u>, the Auction (if necessary) shall have occurred pursuant to the Bidding Procedures Order;
- (vii) No later than the earlier of (a) <u>5 days after any auction</u>, and (b) <u>45 days after the Petition Date</u>, Sellers shall have obtained entry of the Sale Order, the form and substance of which shall be acceptable to Buyer, which approves the Transaction; and
 - (viii) No later than the Outside Date, the Closing shall have occurred.
- (e) From and after the date hereof, Sellers shall not take any action that is intended to result in, or fail to take any action the intent of which failure to act would result in, the reversal, voiding, modification or staying of the Bidding Procedures Order or, if Buyer is the Successful Bidder at the Auction, the Sale Order.

- (f) Sellers further covenant and agree that, after the entry of the Sale Order, the terms of any chapter 11 plan, motion(s) to dismiss or convert the Chapter 11 Cases, or other case closing procedural relief requested that Sellers submit to the Bankruptcy Court, or any other court for confirmation or sanction, shall not be intended to (or reasonably likely to) supersede, abrogate, nullify or restrict the terms of this Agreement in any material respect, or prevent the consummation or performance of the Transactions.
- (g) Sellers acknowledge and agree that Buyer has expended considerable time and expense in connection with this Agreement, and the negotiation thereof, and the identification and quantification of assets to be included in the Acquired Assets. In consideration therefor, the Bidding Procedures Motion shall include a request from Sellers that the Bankruptcy Court approve the Break-Up Fee and Expense Reimbursement as allowed superpriority administrative expense claims against Sellers' estates pursuant to sections 105(a), 503(b), and 507(a)(2) of the Bankruptcy Code with priority over all other administrative expenses of the kind specified in section 503(b) of the Bankruptcy Code.
- Cure Costs. Sellers shall sell, transfer and assign, all Assumed Contracts to Buyer, 6.8 and Buyer shall purchase and assume all Assumed Contracts from Sellers, as of the Closing Date pursuant to sections 363 and 365 of the Bankruptcy Code and the Sale Order. In connection with debts incurred or the assignment and assumption of the Assumed Contracts, Buyer shall cure any monetary defaults of the debts incurred or under the Assumed Contracts by payment of any Cure Costs as determined in accordance with the Bidding Procedures Order and Sale Order. Solicitation of Alternative Transactions in Accordance with Bidding Procedures. From and after the Petition Date, the Sellers and their Representatives shall be permitted to solicit, initiate, encourage, respond to, or take any other action designed to solicit an Alternative Transaction, provided that the foregoing is in accordance with, and limited by, the terms and conditions of the proposed Bidding Procedures, including the Qualified Bid requirements set forth in the proposed Bidding Procedures. If an Auction is conducted, and Buyer is not the Successful Bidder, Buyer shall, in accordance with and subject to the Bidding Procedures, be required to serve as the back-up bidder if Buyer is the next highest or otherwise best bidder at the Auction (such party that is the next highest or otherwise best bidder at the Auction, the "Back-Up Bidder") and, if Buyer is the Back-Up Bidder, Buyer shall, notwithstanding Section 8.1(b)(ii), be required to keep its bid to consummate the Transactions on the terms and conditions set forth in this Agreement (as the same may be improved upon by Buyer in the Auction) open and irrevocable until the Outside Date. Following the Auction, if the Successful Bidder fails to consummate the applicable Alternative Transaction as a result of a breach or failure to perform on the part of such Successful Bidder, then Buyer, if Buyer is the Back-Up Bidder, will be deemed to have the new prevailing bid, and Sellers may seek authority to consummate the Transactions on the terms and conditions set forth in this Agreement (as the same may be improved upon by Buyer in the Auction); provided that the closing shall occur no later than the date that is three (3) Business Days after the Outside Date, unless such date is otherwise extended by Buyer in its sole discretion. Notification of Certain Matters. To the extent permitted by applicable Law, Sellers will give prompt written notice to the Buyer of (a) the existence of any fact or circumstance, or the occurrence of any event, of which it has Knowledge that would reasonably be likely to cause a condition to the Seller's obligations to consummate the Transactions set forth in Article 7 not to be satisfied as of any date, including the breach of any

representation, warranty, or covenant in this Agreement, (b) any notice or other communication from any Person asserting that the consent of such Person which is or may be required in connection with the Transactions is not likely to be obtained prior to Closing or any written objection or proceeding challenging the Transactions or the entry of the approval by the Bankruptcy Court, or (c) the receipt of any notice or other communication from any Governmental Entity in connection with the Transactions contemplated by this Agreement; provided, however, that the delivery of any such notice pursuant to this Section 6.10 shall not be deemed to amend or supplement this Agreement and the failure to deliver any such notice shall not constitute a waiver of any right or condition to the consummation of the Transactions contemplated hereby by any Party. Subject to (i) any confidentiality provisions binding the Sellers and (ii) restrictions of access to classified information, Sellers shall permit Buyer or its Representatives to review in advance any proposed material written or material oral communication or information submitted to any Governmental Entity in connection with the Transactions, shall furnish Buyer with copies of all correspondence, filings and communications with any Governmental Entity, and shall not agree to participate in any meeting with any Governmental Entity related to the Transactions unless Sellers consult with Buyer in advance and, to the extent permitted by any such Governmental Entity, give Buyer the opportunity to attend and participate in such meeting, in each case to the maximum extent practicable and in accordance with applicable Law. Each Party shall furnish the other Party with such necessary information and assistance as such other Party may reasonably request in connection with their preparation of filings, registrations, requests or submissions of information to any Governmental Entity in connection with this Agreement and the Transactions.

- 6.11 <u>Confidentiality</u>. The Parties hereby acknowledge and agree that the Confidentiality Agreement is enforceable in accordance with its terms, it being understood and agreed, however, that such Confidentiality Agreement shall terminate automatically effective as of the Closing.
- Approvals and Consents. Subject to the terms and conditions of this Agreement (including Section 6.14 (Permits)), each Party shall use commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Law to consummate the Transactions, including (i) preparing and filing or otherwise providing, in consultation with the other Party and as promptly as practicable and advisable after the date hereof, all documentation to effect all necessary applications, notices, petitions, filings, and other documents and to obtain as promptly as practicable all waiting period expirations or terminations, consents, clearances, waivers, licenses, orders, registrations, approvals, permits, and authorizations necessary or advisable to be obtained from any third party or any Governmental Entity in order to consummate the Transactions, and (ii) taking all steps as may be necessary, subject to the limitations in this Section 6.12, to obtain all such waiting period expirations or terminations, consents, clearances, waivers, licenses, registrations, permits, authorizations, orders and approvals from any third party or any Governmental Entity. Buyer shall be responsible for all filing fees payable in connection with such filings; provided that all other costs and expenses incurred in connection with this Section 6.12 (including, for the avoidance of doubt, all costs and expenses payable by such Party to any of its financial advisors, attorneys, accountants, advisors, consultants or other Representatives) shall be paid by the Party incurring such costs and expenses.

- 6.13 <u>Update of Disclosure Schedules</u>. From the date of this Agreement until the Closing Date, Sellers shall as promptly as reasonably practicable deliver any new schedules or supplement or amend the Disclosure Schedules with respect to any matter that, if existing, occurring or known after the date of this Agreement, would have been required to be set forth or described in the Disclosure Schedules if such matter had existed, occurred, or been known as of the date of this Agreement.
- 6.14 <u>Permits</u>. As promptly as practicable after the Closing, Sellers and Buyer shall file with each applicable Governmental Entity the notices and information required pursuant to any applicable Law to transfer the Permits included in the Acquired Assets from Sellers to Buyer. The Parties also agree to use all reasonable best efforts to take any and all other actions at Sellers' sole cost and expense required by any Governmental Entity to effect the transfer of the Transferred Permits from Sellers to Buyer. Buyer and Sellers shall jointly be responsible for all filing fees and reasonable out-of-pocket transfer costs payable in connection with this Section 6.14; provided that all other costs and expenses incurred in connection with this Section 6.14 (including, for the avoidance of doubt, all costs and expenses payable by such Party to any of its financial advisors, attorneys, accountants, advisors, consultants or other Representatives) shall be paid by the Party incurring such costs and expenses. Bulk Transfer Laws. Sellers shall ensure that the Sale Order provides either that: (i) Sellers have complied with any applicable bulk sale or bulk transfer Laws of any jurisdiction, including the United Nations Convention on the Sale of Goods, in connection with the Transactions contemplated by this Agreement; or (ii) compliance with such Laws described in the foregoing clause (i) is not necessary or appropriate under the circumstances. Further, the Parties intend that pursuant to section 363(f) of the Bankruptcy Code and to the extent provided for in the Sale Order, the transfer of the Acquired Assets shall be free and clear of any Liens on the Acquired Assets (except Permitted Liens, if any), including any Liens arising out of the bulk transfer Laws, and Sellers shall take such steps as may be necessary or appropriate to so provide in the Sale Order.
- 6.16 <u>Buyer Deficiency Claim</u>. Subject to the consummation of the Transactions contemplated under this Agreement, Sellers acknowledge and agree that Buyer shall be entitled to a deficiency claim, calculated as set forth in <u>Section 3.2(c)</u>. Buyer and Seller shall cooperate in good faith to determine the full quantum of Buyer's secured Claim against Sellers and their chapter 11 estates. The reconciled amount of the Buyer's secured Claim for the purpose of its final Credit Bid Consideration shall be set forth in the Bidding Procedures Order. For the avoidance of doubt, the Bidding Procedures Order shall contain a provision whereby Sellers acknowledge, admit, stipulate, and agree to the validity, quantum, enforceability, and relative priority of the Buyer's reconciled, secured Claims.
- 6.17 <u>Preservation of Records</u>. After the Closing Date, Buyer shall provide to Sellers and their respective Affiliates and Representatives—after reasonable notice and during normal business hours and at Sellers' expense—reasonable access to, including the right to make copies of, all Acquired Business Information included in and otherwise related to the Acquired Assets for periods prior to the Closing and shall preserve such Acquired Business Information until the later of (i) the conclusion of all bankruptcy proceedings relating to the Chapter 11 Cases, or (ii) in the

case of Acquired Business Information relating to Taxes, the expiration of the statute of limitations applicable to such Taxes.

Third-Party Consents. To the extent that Sellers' rights under any Contract or 6.18 Permit constituting an Acquired Asset, or any other Acquired Asset, might not be assigned to Buyer without the consent of another Person that has not been obtained as of the Closing, this Agreement shall not constitute an agreement to assign the same if an attempted assignment would constitute a breach thereof or be unlawful, and the Sellers shall use commercially reasonable efforts to obtain any such required consent(s) as promptly as possible. If any such consent shall not be obtained, or if any attempted assignment would be ineffective or would impair Buyer's rights under the Acquired Asset in question, so that Buyer would not in effect acquire the benefit of all such rights, Sellers, to the maximum extent permitted by applicable Law and the Acquired Asset, shall act after the Closing as Buyer's agent in order to obtain for it the benefits thereunder and shall cooperate, to the maximum extent permitted by applicable Law and the Acquired Asset, with Buyer in any other reasonable arrangement designed to provide such benefits (or equivalent benefits) to Buyer. Furthermore, to the extent that Sellers have not obtained (i) one or more required consents, modifications, or terminations set forth in Section 4.3 of the Disclosure Schedules, and Buyer has not expressly waived satisfaction of such condition in writing, or (ii) any other notice, consent or modification reasonably required or requested by the Buyer at or following the Closing, Sellers shall use commercially reasonable efforts to obtain any such required consents, other notices, consents, modifications, or terminations as promptly as possible following the Closing.

ARTICLE VII. CONDITIONS TO OBLIGATIONS OF THE PARTIES

- 7.1 <u>Conditions Precedent to Obligations of Buyer</u>. The obligation of Buyer to consummate the Transactions is subject to the satisfaction (or written waiver by Buyer) at or prior to the Closing Date in all material respects of each of the following conditions:
- (a) Accuracy of Representations and Warranties. The representations and warranties of Sellers set forth herein, including Article 4 hereof, shall be true and correct in all material respects as of the Agreement Date and at and as of the Closing as though made at and as of the Closing (in each case, except to the extent expressly made as of another date, in which case as of such date as if made at and as of such date).
- (b) <u>Performance of Obligations</u>. Sellers shall have performed all obligations and agreements contained in this Agreement required to be performed by them on or prior to the Closing Date, including all of their performance obligations under <u>Section 3.1(b)</u> hereof.
- (c) <u>Sale Order</u>. After notice and a hearing as defined in Section 102(1) of the Bankruptcy Code, the Bankruptcy Court shall have entered the Sale Order, and such Sale Order (i) shall have become final and non-appealable, (ii) shall not have been reversed, stayed, modified or amended, vacated, and as to which the time to appeal or seek certiorari or move for a vacatur, new trial, reargument or rehearing has expired, and no appeal or petition for certiorari or other proceedings for a vacatur, new trial, reargument or rehearing has been timely taken, or as to which

any appeal that has been taken or any petition for certiorari that has been timely filed has been withdrawn or resolved by the highest court to which the Order was appealed or from which certiorari was sought or the vacatur, new trial, reargument or rehearing shall have been denied or resulted in no modification of such Order, and (iii) shall not have been amended, supplemented or otherwise modified in a manner that results in such Sale Order no longer being in form and substance acceptable to Buyer in its sole discretion.

- (d) <u>Officer's Certificate</u>. Buyer shall have received a certificate, dated the Closing Date, of an executive officer of each Seller to the effect that the conditions specified in <u>Section 7.1(a)</u>, and <u>Section 7.1(b)</u> above have been fulfilled.
- (e) <u>No Order</u>. No court or other Governmental Entity has issued, enacted, entered, promulgated or enforced any Law or Order (that is final and non-appealable and that has not been vacated, withdrawn or overturned) restraining, staying, enjoining or otherwise prohibiting the Transactions.
- (f) <u>Compliance with Indication of Interest</u>. Sellers shall have fully performed those obligations required to be performed prior to the Agreement Date under that certain Indication of Interest, dated as of March 31, 2023, by and between Sellers and Buyer (the "<u>IOI</u>").
- (g) <u>Governmental Approvals</u>. (i) All approvals or consents of Governmental Entities required prior to Closing that are material to the Business or to Buyer and its Affiliates shall have been obtained, and (ii) there shall be no Law or Order that restrains or prevents the Transactions.
- (h) <u>Credit Bid</u>. (i) The Bankruptcy Court shall have entered a final non-appealable Order permitting Buyer to offset not less than the amount of the Credit Bid Consideration, of Buyer's then-outstanding secured Claims as of the Closing, and (ii) there shall be no cause under section 363(k) of the Bankruptcy Code prohibiting Buyer from offsetting not less than the amount of the Credit Bid Consideration, of Buyer's outstanding secured Claims at the time of Closing as consideration for the Acquired Assets.
- (i) <u>Assumed Contracts; Consents</u>. To the extent the assignment of an Assumed Contract is not authorized by the Sale Order, Sellers shall have obtained any consent or approval required under the Assumed Contracts as a result of the execution or performance of this Agreement or the consummation of the Transactions, in form and substance reasonably satisfactory to Buyer, and each such consent shall be in full force and effect and shall not have been revoked.
- (j) <u>Permits</u>. Sellers shall have assigned to Buyer pursuant to Section 365 of the Bankruptcy Code and the Sale Order, any Permits that Buyer determines in its reasonable discretion to be material to the Business, and such Permits shall be in full force and effect and shall not have been revoked.

- 7.2 <u>Conditions Precedent to the Obligations of Sellers</u>. The obligation of Sellers to consummate the Transactions is subject to the satisfaction (or written waiver by Sellers) at or prior to the Closing Date in all material respects of each of the following conditions:
- (a) <u>Accuracy of Representations and Warranties</u>. The representations and warranties of Buyer set forth herein, including <u>Article 5</u> hereof, shall be true and correct in all material respects as of the Agreement Date and at and as of the Closing as though made at and as of the Closing (in each case, except to the extent expressly made as of another date, in which case as of such date as if made at and as of such date).
- (b) <u>Performance of Obligations</u>. Buyer shall have performed in all material respects all obligations and agreements contained in this Agreement required to be performed by it on or prior to the Closing Date.
- (c) <u>Sale Order</u>. After notice and a hearing as defined in Section 102(1) of the Bankruptcy Code, the Bankruptcy Court shall have entered the Sale Order, and such Sale Order (i) shall have become final and non-appealable, (ii) shall not have been reversed, stayed, modified or amended, vacated, and as to which the time to appeal or seek certiorari or move for a vacatur, new trial, reargument or rehearing has expired, and no appeal or petition for certiorari or other proceedings for a vacatur, new trial, reargument or rehearing has been timely taken, or as to which any appeal that has been taken or any petition for certiorari that has been timely filed has been withdrawn or resolved by the highest court to which the Order was appealed or from which certiorari was sought or the vacatur, new trial, reargument or rehearing shall have been denied or resulted in no modification of such Order, and (iii) shall not have been amended, supplemented or otherwise modified in a manner that results in such Sale Order no longer being in form and substance acceptable to Buyer in its sole discretion.
- (d) <u>Officer's Certificate</u>. Sellers shall have received a certificate, dated the Closing Date, of an executive officer of Buyer to the effect that the conditions specified in Section 7.2(a) and Section 7.2(b) above have been fulfilled.
- (e) <u>No Order</u>. No court or other Governmental Entity has issued, enacted, entered, promulgated or enforced any Law or Order (that is final and non-appealable and that has not been vacated, withdrawn or overturned) restraining, staying, enjoining or otherwise prohibiting the Transactions.
- 7.3 <u>Frustration of Conditions Precedent</u>. Neither Buyer nor Sellers may rely on the failure of any condition set forth in this <u>Article 7</u>, as applicable, to be satisfied if such failure was caused by such Party's breach of this Agreement.

ARTICLE VIII. TERMINATION

8.1 <u>Termination of Agreement</u>. This Agreement may be terminated and the Transactions abandoned at any time prior to the Closing:

- (a) by mutual written agreement of Sellers and Buyer;
- (b) by either Sellers or Buyer:
- (i) if there shall be any Law that makes consummation of the Transactions illegal or otherwise prohibited, or if any Order permanently restraining, prohibiting, staying or enjoining Buyer or Sellers from consummating the Transactions is entered and such Order shall become final; <u>provided</u>, <u>however</u>, that no termination may be made by a Party under this <u>Section 8.1(b)(i)</u> if the issuance of such Order was caused by the breach of any representations, warranties, covenants or agreements contained in this Agreement by such Party;
- (ii) subject to <u>Section Error! Reference source not found.</u>, upon the Bankruptcy Court's entry of an order approving an Alternative Transaction;
- (iii) if the Closing shall not have occurred on or before the Outside Date, <u>provided</u>, <u>however</u> that no termination may be made by a Party under this <u>Section 8.1(b)(iii)</u> if the failure to close on or before the Outside Date was caused by the breach of any representations, warranties, covenants or agreements contained in this Agreement by such Party;
- (iv) if (A) a trustee or an examiner with expanded powers is appointed in any of the Chapter 11 Cases, or (B) any of the Chapter 11 Cases are dismissed or converted to a case under Chapter 7 of the Bankruptcy Code;
- (c) by Buyer if there has been a material breach by any Seller of any representation, warranty, covenant, or agreement contained in this Agreement that would prevent the satisfaction of the conditions to the obligations of Buyer at Closing set forth in Section 7.1(a) and Section 7.1(b), and such breach has not been waived by Buyer, or, if such breach is curable, cured by such Seller prior to the earlier to occur of (A) five (5) days after receipt of Buyer's notice of such breach, and (B) the Outside Date.
- (d) by Buyer if, subject to <u>Section Error! Reference source not found.</u>, (i) Sellers enter into an Alternative Transaction, or (ii) Buyer is neither the Successful Bidder nor the Back-Up Bidder following an Auction.
- (e) by Buyer if (i) any Milestone is not satisfied for any reason, (ii) any payment obligation set forth in Section 6.1(b) is not satisfied when due for any reason, or (iii) at any time after entry of the Bidding Procedures Order or the Sale Order, such Orders are reversed, stayed for more than fourteen (14) days, vacated or modified.
- (f) by the Buyer if for any reason Buyer is not authorized, pursuant to section 363(k) of the Bankruptcy Code or otherwise, to credit bid not less than the Credit Bid Consideration, and up to \$7,588,255.00, in payment of the Purchase Price as set forth in Section 3.2(a) and the reconciliation set forth in Section 6.16.
- (g) by Buyer if any holders of Sellers' Series A Preferred Stock or Series B Preferred Stock file a document in any court, or otherwise publicly make an assertion, contending

that a Seller does not have authority or the requisite consents to enter into this Agreement, commence the Chapter 11 Cases, or otherwise consummate the Transactions contemplated herein, unless such filing is withdrawn or overruled by a court of competent jurisdiction on a final basis with prejudice or such assertion is publicly revoked, and, for the avoidance of doubt, notwithstanding any disclosures made by Sellers in Section 4.4 of the Disclosure Schedules.

- (h) by Sellers if there has been a material breach by any Buyer of any representation, warranty, covenant, or agreement contained in this Agreement that would prevent the satisfaction of the conditions to the obligations of Sellers at Closing set forth in Section 7.2(a) and Section 7.2(b), and such breach has not been waived by Sellers, or, if such breach is curable, cured by Buyer prior to the earlier to occur of (A) five (5) days after receipt of Sellers' notice of such breach, and (B) the Outside Date; provided, that Sellers shall not have a right of termination pursuant to this Section 8.1(h) if Buyer could, at such time, terminate this Agreement pursuant to Section 8.1(c).
- (i) by Sellers if the governing body of Sellers determines, upon advice from outside legal counsel, that failing to terminate this Agreement would violate its or such governing body's fiduciary obligations, including to consummate an Alternative Transaction.

8.2 <u>Consequences of Termination</u>.

(a) In the event of any termination of this Agreement by either or both of Buyer and Sellers pursuant to Section 8.1, written Notice thereof shall be given by the terminating Party to the other Parties hereto, specifying the provision hereof pursuant to which such termination is made, this Agreement shall thereupon terminate and become void and of no further force and effect, except for this Section 8.2 (Consequences of Termination), and Section 3.3 (Deposit), Section 6.1(b) (Interim Operating Covenants), Section 6.3 (Public Announcements), Section 6.11 (Confidentiality.), Article 9 (Miscellaneous), and to the extent applicable in respect of such Sections and Article, Article 1 (Definitions)), and the Transactions shall be abandoned without further action of the Parties hereto, except that, subject to Section 8.2(c) such termination shall not relieve any Party of any Liability for Fraud, gross negligence, or breach of this Agreement.

(c) <u>Expense Reimbursement and Break-Up Fee</u>.

- (i) Sellers shall, within two (2) Business Days after any termination of this Agreement other than pursuant to <u>Section 8.1(h)</u>, pay to Buyer, by wire transfer of immediately available funds, cash in an amount equal to the Expense Reimbursement.
- (ii) If this Agreement is terminated for any reason other than pursuant to Section 8.1(h), Sellers shall pay to Buyer, by wire transfer of immediately available funds, cash in an amount equal to the Break-Up Fee upon the consummation of an Alternative Transaction. Any payment made pursuant to this Section 8.2(b)(ii) shall be made at the closing of the Alternative Transaction and shall be payable from the proceeds of such Alternative Transaction.
- (iii) Sellers acknowledge and agree that (A) the payment of the Expense Reimbursement and Break-Up Fee are integral parts of the Transactions, (B) in the absence of

Sellers' obligations to make these payments, Buyer would not have entered into this Agreement, (C) time is of the essence with respect to the payment of the Expense Reimbursement and Break-Up Fee, and (D) the Expense Reimbursement and Break-Up Fee shall constitute allowed superpriority administrative expense claims against Sellers' estates pursuant to sections 105(a), 503(b), and 507(a)(2) of the Bankruptcy Code with priority over all other administrative expenses of the kind specified in section 503(b) of the Bankruptcy Code.

ARTICLE IX. MISCELLANEOUS

- 9.1 <u>Expenses</u>. Except as set forth in this Agreement and whether or not the Transactions are consummated, each Party hereto shall bear all costs and expenses incurred or to be incurred by such Party in connection with this Agreement and the consummation of the Transactions.
- 9.2 <u>Assignment</u>. Neither this Agreement nor any of the rights or obligations hereunder may be assigned by Sellers without the prior written consent of Buyer, or by Buyer without the prior written consent of Sellers; <u>provided</u>, that Buyer may, without the consent of Sellers, assign this Agreement and its rights and obligations hereunder in whole or in part to any Affiliate of Buyer. Subject to the foregoing, this Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and permitted assigns, including any liquidating trustee, responsible Person, or similar representative for Sellers or Sellers' estate appointed in connection with the Chapter 11 Cases.
- 9.3 <u>Parties in Interest</u>. This Agreement shall be binding upon and inure solely to the benefit of Sellers (and their estates), Buyer and their respective successors or permitted assigns. Nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any rights, benefits or remedies of any nature whatsoever under or by reason of this Agreement except as expressly set forth herein.
- 9.4 <u>Notices</u>. All notices, demands, requests, consents, approvals or other communications (collectively, "<u>Notices</u>") required or permitted to be given hereunder or that are given with respect to this Agreement shall be in writing and shall be personally served, delivered by a nationally recognized overnight delivery service with charges prepaid, or transmitted by hand delivery, or facsimile or electronic mail, addressed as set forth below, or to such other address as such Party shall have specified most recently by written Notice. Notice shall be deemed given on the date of service or transmission if personally served or transmitted by facsimile or electronic mail with confirmation of receipt; <u>provided</u> that if delivered or transmitted on a day other than a Business Day or after 5:00 p.m. Eastern Time, notice shall be deemed given on the next Business Day. Notice otherwise sent as provided herein shall be deemed given on the next Business Day following timely deposit of such Notice with an overnight delivery service:

If to any Seller: PLx Pharma Inc.

PLx Opco Inc.

9 Fishers Lane, Suite E Sparta, New Jersey 07871

Attention: Natasha Giordano

Email: ngiordano@plxpharma.com

With copies to: Raymond James & Associates

320 Park Avenue

New York, New York 10022

Attention: Raymond James, Investment Banker Email: Geoffrey.Richards@RaymondJames.com

-and-

Olshan Frome Wolosky LLP 1325 Avenue of the Americas New York, New York 10019 Attention: Adam H. Friedman

Email: afriedman@olshanlaw.com

If to Buyer: PLx Acquisition Company, LLC

c/o Greenwood Brands LLC

4455 Genesee Street Buffalo, New York 14225

Attention: George Cleary, Authorized Signatory

Email: george.cleary@mpgllc.com

With copies to: DLA Piper LLP (US)

444 West Lake Street, Suite 900

Chicago, Illinois 60606

Attention: W. Benjamin Winger and Katie Allison Email: benjamin.winger@us.dlapiper.com

katie.allison@us.dlapiper.com

Rejection of or refusal to accept any Notice, or the inability to deliver any Notice because of changed address of which no Notice was given, shall be deemed to be receipt of the Notice as of the date of such rejection, refusal or inability to deliver.

9.5 <u>Choice of Law</u>. Except to the extent the mandatory provisions of the Bankruptcy Code apply, this Agreement shall be construed and interpreted, and the rights of the Parties shall be determined, in accordance with the Laws of the State of Delaware, without giving effect to any provision thereof that would require or permit the application of the substantive laws of any other jurisdiction.

- 9.6 Entire Agreement; Amendments and Waivers. This Agreement, the Sale Order, and all Transaction Documents and all certificates and instruments delivered pursuant hereto and thereto constitute the entire agreement among the Parties pertaining to the subject matter hereof and supersede all prior agreements, understandings, negotiations, and discussions, whether oral or written, of the Parties. This Agreement may be amended, supplemented or modified, and any of the terms, covenants, representations, warranties or conditions may be waived, only by a written instrument executed by Buyer and Sellers, or in the case of a waiver, by the Party waiving compliance. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar), and no such waiver shall constitute a continuing waiver unless otherwise expressly provided.
- 9.7 <u>Counterparts; Facsimile and Electronic Signatures</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, *e.g.*, www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes. In proving this Agreement, it shall not be necessary to produce or account for more than one such counterpart signed by the Party against whom enforcement is sought.
- 9.8 <u>Severability</u>. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provisions of this Agreement. In the event that any of the provisions of this Agreement shall be held by a court or other tribunal of competent jurisdiction to be illegal, invalid or unenforceable, such provisions shall be limited or eliminated only to the minimum extent necessary so that this Agreement shall otherwise remain in full force and effect.
- 9.9 <u>Headings</u>. The table of contents and the headings of the Articles and Sections herein are inserted for convenience of reference only and are not intended to be a part of, or to affect the meaning or interpretation of, this Agreement.

9.10 <u>Exclusive Jurisdiction; Specific Performance</u>.

- (a) Subject to Section 9.10(b), without limiting any Party's right to appeal any order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to enforce the terms of this Agreement and to decide any Claims or disputes which may arise or result from, or be connected with, this Agreement, any breach or default hereunder, or the Transactions, and (ii) any and all Proceedings related to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the Parties hereby consent to and submit to the jurisdiction and venue of the Bankruptcy Court and shall receive Notices at such locations as indicated in Section 9.4. For the avoidance of doubt, this Section 9.10 shall not apply to any Claims that Buyer or its Affiliates may have against any third party following the Closing.
- (b) Notwithstanding anything herein to the contrary, in the event the Chapter 11 Cases of Sellers are closed or dismissed, the Parties hereby agree that all Claims or disputes

which may arise or result from, or be connected with, this Agreement, any breach or default hereunder, or the Transactions, shall be heard and determined exclusively in any federal court sitting in the District of Delaware or, if that court does not have subject matter jurisdiction, in any state court located in the County of New Castle, Delaware (and, in each case, any appellate court thereof), and the Parties hereby consent to and submit to the jurisdiction and venue of such courts.

- (c) Buyer acknowledges that Sellers would be damaged irreparably in the event that the terms of this Agreement are not performed by Buyer in accordance with its specific terms or otherwise breached or Buyer fails to consummate the Closing and that, in addition to any other remedy that Sellers may have under law or equity, Sellers shall be entitled to seek injunctive relief to prevent breaches of the terms of this Agreement and to seek to enforce specifically the terms and provisions hereof that are required to be performed by Buyer. Buyer further agrees that no Seller shall be required to obtain, furnish or post any bond or similar instrument in connection with or as a condition to obtaining any remedy referred to in this Section 9.10, and irrevocably waives any right it may have to require the obtaining, furnishing or posting of any such bond or similar instrument.
- (d) Sellers acknowledge that Buyer would be damaged irreparably in the event that the terms of this Agreement are not performed by Sellers in accordance with its specific terms or otherwise breached or Sellers fail to consummate the Closing and that, in addition to any other remedy that Buyer may have under law or equity, Buyer shall be entitled to seek injunctive relief to prevent breaches of the terms of this Agreement and to seek to enforce specifically the terms and provisions hereof that are required to be performed by Sellers. Each Seller further agrees that Buyer shall not be required to obtain, furnish or post any bond or similar instrument in connection with or as a condition to obtaining any remedy referred to in this Section 9.10, and irrevocably waives any right it may have to require the obtaining, furnishing or posting of any such bond or similar instrument.
- 9.11 <u>WAIVER OF RIGHT TO TRIAL BY JURY</u>. SELLERS AND BUYER HEREBY WAIVE TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT THEY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). FOR THE AVOIDANCE OF DOUBT, THIS <u>SECTION 9.11</u> SHALL NOT APPLY TO ANY CLAIMS THAT BUYER OR ITS AFFILIATES MAY HAVE AGAINST ANY THIRD PARTY FOLLOWING THE CLOSING.
- 9.12 <u>Survival</u>. Each and every representation and warranty contained in this Agreement shall expire and be of no further force and effect as of the Closing. Each and every covenant and agreement contained in this Agreement (other than the covenants contained in this Agreement which by their terms are to be performed (in whole or in part) by the Parties following the Closing (each, a "<u>Post-Closing Covenant</u>")) shall expire and be of no further force and effect as of the Closing. Each Post-Closing Covenant shall survive the Closing until the earlier of (a) performance of such Post-Closing Covenant in accordance with this Agreement or, (b) (i) if time for performance of such Post-Closing Covenant is specified in this Agreement, sixty (60) days following the expiration of the time period for such performance or (ii) if time for performance of

such Post-Closing Covenant is not specified in this Agreement, the expiration of the applicable statute of limitations with respect to any claim for any failure to perform such Post-Closing Covenant; provided that if a written notice of any claim with respect to any Post-Closing Covenant is given prior to the expiration thereof then such Post-Closing Covenant shall survive until, but only for purposes of, the resolution of such claim by final, non-appealable judgment or settlement.

- Non-Recourse. All Claims or causes of action (whether in contract or in tort, in law or in equity, or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with, or related in any manner to this Agreement or the Transaction Documents may be made only against (and are expressly limited to) the Persons that are expressly identified as parties hereto or thereto (the "Contracting Parties"). In no event shall any Contracting Party have any shared or vicarious Liability for the actions or omissions of any other Person. No Person who is not a Contracting Party, including any director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney or representative of, and any financial advisor or lender to, any of the foregoing ("Non-Party Affiliates"), shall have any Liability (whether in contract or in tort, in law or in equity, or granted by statute or based upon any theory that seeks to impose Liability of an entity party against its owners or Affiliates) for any claims, causes of action, obligations or Liabilities arising under, out of, in connection with or related in any manner to this Agreement or the Transaction Documents or based on, in respect of, or by reason of this Agreement or the Transaction Documents or their negotiation, execution, performance or breach; and, to the maximum extent permitted by Law, each Contracting Party hereby knowingly and voluntarily waives and releases all such Liabilities, claims and obligations against any such Non-Party Affiliates. Without limiting the foregoing, to the maximum extent permitted by Law, (a) each Contracting Party hereby knowingly and voluntarily waives and releases any and all rights, claims, demands, or causes of action that may otherwise be available at law or in equity, or granted by statute, to avoid or disregard the entity form of a Contracting Party or otherwise impose Liability of a Contracting Party on any Non-Party Affiliate, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise; and (b) each Contracting Party disclaims any reliance upon any Non-Party Affiliates with respect to the performance of this Agreement or the Transaction Documents or any representation or warranty made in, in connection with, or as an inducement to this Agreement or the Transaction Documents.
- 9.14 <u>Disclosure Schedules</u>. Except as set forth in this Agreement, the inclusion of any information (including dollar amounts) in the Disclosure Schedules shall not be deemed to be an admission or acknowledgment by any Party that such information is required to be listed on such section of the relevant schedule or is material to or outside the Ordinary Course of Business of any Person. The information contained in this Agreement, the exhibits hereto and the Disclosure Schedules, is disclosed solely for purposes of this Agreement, and no information contained herein or therein shall be deemed to be an admission by any Party to any third party of any matter whatsoever (including any violation of any Law or breach of contract). Unless the context otherwise requires, all capitalized terms used in the Disclosure Schedules shall have the respective meanings assigned in this Agreement. The Disclosure Schedules set forth items of disclosure with specific reference to the particular Section or subsection of this Agreement to which the

information in the Disclosure Schedules relates; <u>provided</u>, <u>however</u>, that any information set forth in one Section of the Disclosure Schedules will be deemed to apply to each other section or subsection thereof to which its relevance is reasonably apparent on its face.

- 9.15 <u>Mutual Drafting</u>. This Agreement is the result of the joint efforts of Buyer and Sellers, and each provision hereof has been subject to the mutual consultation, negotiation and agreement of the Parties and there is to be no construction against any Party based on any presumption of that Party's involvement in the drafting thereof.
 - 9.16 <u>Time of Essence</u>. Time is of the essence of this Agreement.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of Sellers and Buyer as of the date first above written.

BUYER:

PLx Acquisition Company, LLC

By: George Cleary

Name: George Cleary

Title: Authorized Signatory

---- DocuSigned by:

REEL: 064913 FRAME: 0454

SELLERS:

PLx Pharma Inc.

Name: Natasha Giordano

Title: President and Chief Executive Officer

PLx Opco Inc.

By: Name: Natasha Giordano

Title: President and Chief Executive Officer

AMENDMENT NO. 1 TO ASSET PURCHASE AGREEMENT

This Amendment No. 1 (this "<u>Amendment</u>") to the Asset Purchase Agreement (the "<u>Agreement</u>") entered into and effective as of April 12, 2023, is entered into as of May [●], 2023, by and among PLx Pharma Inc., a Delaware corporation (the "<u>Company</u>") and PLx Opco Inc., a Delaware corporation and a wholly-owned subsidiary of the Company (together with the Company, "<u>Sellers</u>" and each, a "<u>Seller</u>"), and PLx Acquisition Company, LLC, a Delaware limited liability company and a wholly-owned subsidiary of Greenwood ("<u>Buyer</u>"). Each Seller and Buyer are referred to herein individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>". All capitalized terms used but not otherwise defined in this Amendment shall have the meaning ascribed in the Agreement.

WHEREAS, following further discussion and analysis, the Parties have mutually agreed to revise certain terms of the Agreement; and

WHEREAS, the undersigned, constituting each Party to the Agreement, wish to amend the Agreement in accordance with <u>Section 9.6</u> of the Agreement, in the manner set forth herein.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties hereto agree as follows:

- 1. **Definition of Acquired Assets**. Effective as of the Effective Date:
- (a) <u>Subsection (d)</u> of the definition of "Acquired Assets" set forth in <u>Section 1.1</u> of the Agreement is hereby amended and restated in its entirety to read as follows:
 - "(d) all Seller Intellectual Property, including (without limitation), those items listed in Section 4.11(a) of the Disclosure Schedules, and for the avoidance of doubt, including Sellers' Amazon Seller Central account;"
- (b) <u>Subsection (i)</u> of the definition of "Acquired Assets" set forth in <u>Section 1.1</u> of the Agreement is hereby amended and restated in its entirety to read as follows:
 - "(i) to the extent transferrable under applicable Law, any Tax credits, Tax deposits, Tax rebates, Tax attributes, and prepaid Tax amounts of the Sellers; except for Tax refunds and net operating losses, which, for the avoidance of doubt, are set forth and further described in <u>Section 1.1(b)</u> of the Disclosure Schedule and defined as "Excluded Assets" hereunder;"
- (c) <u>Subsection (m)</u> of the definition of "Acquired Assets" set forth in <u>Section 1.1</u> of the Agreement is hereby amended and restated in its entirety to read as follows:
 - "(m) to the extent transferrable, all security deposits (other than any security deposits held by Sellers' landlord in connection with Sellers' leased real property

PATENT REEL: 064913 FRAME: 0456 which, for the avoidance of doubt, is defined as an "Excluded Asset" hereunder) and any other deposits held by vendors, trade creditors, or any other party, in each case only to the extent related to the Acquired Assets, the Assumed Liabilities, or the Business (or any portion thereof), other than the deposit maintained by Sellers pursuant to the order approving Sellers' 'first-day' utility motion;"

- 2. **Definition of Excluded Assets**. Effective as of the Effective Date, <u>subsection (k)</u> of the definition of "Excluded Assets" set forth in <u>Section 1.1</u> of the Agreement shall be renamed <u>subsection (l)</u> and a new <u>subsection (k)</u> shall be added to the definition of "Excluded Assets" to read as follows:
 - "(k) all security deposits held by Sellers' landlord in connection with Sellers' leased real property; and"
- 3. **Definition of Inventory**. Effective as of the Effective Date, the definition of "Inventory" set forth in <u>Section 1.1</u> of the Agreement is hereby amended and restated in its entirety to read as follows:
 - "Inventory' shall mean all raw materials, active pharmaceutical ingredients, excipients, work-in-process, semi-finished and finished goods and Products, supplies (including clinical drug supplies), samples, components, packaging materials, and other inventories owned by Sellers, excluding the raw materials listed in Section 1.1(d)(i) of the Disclosure Schedules. For the avoidance of doubt, the finished goods owned and held by Patheon Manufacturing Services LLC ("Patheon") and set forth in Section 1.1(d)(ii) of the Disclosure Schedules shall not be deemed "Inventory" owned by Sellers and shall not be defined as an "Acquired Asset" hereunder."
- 4. **Buyer Deficiency Claim**. Effective as of the Effective Date, <u>Section 6.16</u> of the Agreement is hereby amended and restated in its entirety to read as follows:
 - "6.16 [Intentionally Omitted]."

Furthermore, references in the Agreement to "Section 6.16" shall be deleted.

- 5. **Effective Time**. This Amendment shall become effective upon the entry of the Sale Order (the "<u>Effective Time</u>"). If the Sale Order is not entered, this Amendment shall be rendered void.
- 6. **Effect of Amendment**. The Agreement is amended hereby solely as provided in this Amendment and, as so amended, continues in full force and effect.
- 7. **Counterparts**. This Amendment may be executed in one or more counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, *e.g.*,

www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

- 8. **Defined Terms**. Unless otherwise defined herein, the capitalized terms used herein have the meanings ascribed to them in the Agreement.
- 9. **Miscellaneous**. Sections 9.5 (Choice of Law), 9.6 (Entire Agreement; Amendments and Waiver), 9.9 (Headings), 9.12 (Survival) and 9.15 (Mutual Drafting) of the Agreement are hereby incorporated by reference into, and shall apply to, this Amendment, mutatis mutandis.

[Signature page follows]

IN WITNESS WHEREOF, this Amendment has been duly executed and delivered by the duly authorized officers of Sellers and Buyer as of the date first above written.

BUYER:

PLx Acquisition Company, LLC
By: Name: George Cleary
Title: Authorized Signatory
<u>SELLERS</u> :
PLx Pharma Inc.
By:
Name: Natasha Giordano Title: President and Chief Executive Officer
PLx Opco Inc.
By:
Name: Natasha Giordano

Title: President and Chief Executive Officer

PATENT
RECORDED: 09/14/2023 REEL: 064913 FRAME: 0459